Workers' Preferences over Payment Schedules: Evidence from Ridesharing Drivers

Thiago Scarelli University of Oxford thiago.scarelli@economics.ox.ac.uk

Abstract: This paper investigates the importance of quick remuneration for gig workers. To explore this question, I run a large-scale survey experiment with ridesharing drivers in Brazil. The main finding is that the median driver would be willing to forgo a third of their potential earnings to be paid on the same day of their rides, compared to the alternative of being paid a month later. Such a strong preference for quick pay seems to be associated with liquidity constraints, as drivers under heavier financial stress are more likely to prioritize same-day remuneration. I also document that priming drivers to think about their personal budget makes them more inclined to favor larger (instead of faster) payments, suggesting that pay-me-now can be a default choice for this population. These results advance the literature on job attributes by showing that payment timing is a relevant aspect of an occupation. This chapter also contributes to the gig work debate by emphasizing that digital platforms are best positioned to offer agile pay schemes, which help workers address liquidity shortages in the short run but might induce poverty traps over the long run.

JEL codes: D91; J22; J24; J31; M52. **Keywords:** Labor Supply; Platform Work; Gig Economy; Payment Timing; Liquidity Constraints.

This working paper was last updated on 14 January 2025. The author thanks Maria Balgova, Luc Behaghel, Béatrice Boulu-Reshef, Christina Brown, Fiona Burlig, Liza Charroin, Denis Cogneau, Joshua Dean, Johannes Haushofer, Randi Hjalmarsson, Alex Imas, Nicolas Jacquemet, Anett John, Sylvie Lambert, John List, Karen Macours, David Margolis, Simone Moriconi, Suanna Oh, Pieter Serneels, Neil Thakral, Gabriel Ulyssea, Jack Willis, and Liam Wren-Lewis for their comments. The author is also thankful for the suggestions received from the participants of the PSE-CEPR Policy Forum, the Society of Labor Economists Annual Meeting, the International Conference on Development Economics, the European Economic Association and Econometric Society Congress, and the European Association of Labor Economists Annual Meeting. The author is grateful for the operational collaboration with the company in the collection of the data. This work has been funded by a French government subsidy managed by the Agence Nationale de la Recherche under the framework of the program "Investissements d'avenir", reference ANR-17-EURE-001, and by the Université Paris I Panthéon-Sorbonne. The author declares that he has no financial or non-financial competing interests that could have appeared to influence the work reported in this paper. This study is registered in the AEA RCT Registry under AEARCTR-0010331.

1. Introduction

People working by themselves are often paid less than their peers who have wage jobs, and are systematically overrepresented among the poorest workers in their local labor markets.¹ This statistical regularity has gained renewed attention with the recent increase of gig work in its modern form, in which labor services are mediated by digital platforms, which include ridesharing and delivery services.² For policymakers now facing the challenge of regulating these platforms, it is crucial to understand why people take up these activities despite the relatively low pay rates. One potential reason is that workers appreciate these jobs' extra autonomy and flexibility. However, it is still unclear if these non-monetary benefits are enough to compensate for the magnitude of the earnings penalty they suffer.

In this chapter, I propose and investigate another reason why gig work might be attractive: its rapid payment timing. In essence, gig workers are not only able to adjust their working hours as needed, but they are also paid relatively fast for their services. From the workers' perspective, quickly securing some income might be crucial, especially when consumption needs are pressing or there are few liquidity sources available other than one's own labor. Most forms of own-account work — including modern ridesharing and delivery activities can offer this benefit, as their earnings can be cashed in by the workers faster than the 15 or 30-day intervals that are typical for employees.

If this hypothesis is true, we should expect that the workers taking up those occupations would indeed be willing to trade off larger earnings for faster payments. That is the motivation for the key empirical questions this chapter addresses. In practice, how much value do gig workers assign to quick remuneration? Who values this feature the most? Moreover, since this preference is potentially related to liquidity, how does the salience of one's financial conditions at home affect one's priorities when facing this trade-off?

The difficulty lies in the identification of this preference in a real-world setting. The workers that are paid shortly after their services (such as daily construction workers, hairdressers working on their own, street vendors, or ridesharing drivers) are in many ways different from those with longer payment terms (such as office workers with monthly paychecks or consultants paid after a long project). Without imposing further assumptions, it is difficult to isolate the marginal importance of the payment timing just from the distribution of workers and payment schemes.

^{1.} See Gindling and Newhouse (2014), Bandiera et al. (2022), Scarelli (2022), and Scarelli and Margolis (2023).

^{2.} As discussed by Oyer (2020), International Labour Office (2021), and Garin et al. (2023).

This chapter addresses this challenge by exploring the setting of the ridesharing drivers using a survey experiment in the field. The choice to focus on ridesharing here has two advantages. First, this activity is of intrinsic interest to researchers and policymakers since it represents a new form of labor market engagement. Second, from a methodological perspective, this setting is particularly well-suited for the identification of preferences for quick payment, as it combines three advantages: (a) all workers perform a homogeneous, welldefined task, (b) the time to remuneration is a salient feature of the activity, and (c) payment rules can potentially be changed at the platform's discretion without affecting the fundamental nature of the job.

Leveraging this context, I run a discrete choice experiment with over 14,000 drivers who work with a major ridesharing platform in Brazil. The key outcome of interest is the drivers' reported preference when facing a hypothetical comparison between being paid their usual rate per kilometer always on the same day of their rides, or receiving a higher rate always 30 days after their rides. With the manipulation of the pair of rates they chose from, it is possible to identify an interval of forgone compensation that represents the relative importance of the rapid remuneration timing for each individual driver.

The main result from this elicitation protocol is that the median driver would rather be paid the same day than wait 30 days to receive a fare 1.48 times higher. This choice is equivalent to forgoing one-third of one's nominal earnings per unit of effort (0.48 out of 1.48) in exchange for the benefit of being paid faster. In other words, the median compensated willingness to pay (WTP) for same-day remuneration is at least 33%.

What may explain such high levels of WTP? The survey includes a randomized module just before the preference elicitation protocol to uncover some potential mechanisms behind this result. A third of the respondents are asked how they would cover some unexpected expense, another third is asked how they would use some unexpected income of the same magnitude, and the remaining group serves as a control. Such a design provides a large sample of textual descriptions, offering us a rich insight into the drivers' economic life, while exogenously inducing them to mentally retrieve their financial circumstances, a manipulation that identifies the effect of salient household budgets on payment timing preferences.

Taking stock of the results, a strong preference for fast payment (a) reflects a structural context of resource scarcity and liquidity constraints combined with (b) a modest degree of behavioral heuristics that favors quick pay as a default safe choice. The first point is supported by the finding that drivers living in the poorest households tend to have the highest levels of WTP. Text analysis techniques refine this result by highlighting the feedback interaction between resource scarcity and liquidity: the workers who would choose to receive more

are the ones who already have precautionary reserves or could use their credit cards. At the same time, those who prioritize being paid faster tend to rely on family support when facing temporary shocks — or would need to work longer hours to make up for unexpected expenses.

For the second point, an analysis of the experimental treatment shows that the drivers randomly exposed to the budget questions take a few seconds longer to choose their preferred contract and end up assigning marginally lower importance to be paid faster (or, equivalently, higher importance to earn more) relative to the control group. While it would be plausible to expect people to react differently depending on the content of the hypothetical shock they discussed (unexpected expense or unexpected income), the results suggest that it is the introspective financial exercise in itself that affects the workers' reactions to the intertemporal trade-off in focus, since both treatment arms lead to a similar reduction of about 1.5 percentage point in the WTP for same-day remuneration. This effect is coherent with the hypothesis that fast payment is a default choice (as it is preferred more often in the unprimed group), while the later payment requires a more costly cognitive operation involving the management of deferred flows in the context of one's current conditions (which is kick-started by the forced information retrieval from the budget discussion).

The nature of the hypothetical, non-incentivized elicitation mechanism imposes an important limitation on these results. The preferences reported by the drivers will be meaningful proxies of real-life decisions to the extent that the subjects (a) can understand the proposed trade-off, (b) can anticipate what their decision would be, and (c) do not misrepresent their choices. Those assumptions are plausible in my experimental setting because ridesharing drivers are the experts when it comes to reasoning in terms of kilometer fares. Moreover, they can anticipate the actual consequences of the changes in payment rules proposed in the experiment better than the rest of the population, given that their income is a function of the earnings from their rides.

This chapter contributes to four strands of the economic literature. Firstly, it documents that workers can attach very high value to the simple job feature of being paid shortly after the task, extending the debate on job attributes. In this sense, the proposed measurement of the WTP for same-day remuneration is close in spirit to the elicitation of WTP for work flexibility (Mas and Pallais 2017; K.-M. Chen et al. 2020), for less commute time (Le Barbanchon, Rathelot, and Roulet 2021), for stability and earnings growth (Wiswall and Zafar 2018), and for fringe benefits (Eriksson and Kristensen 2014).

Secondly, this research also relates to the extensive literature on intertemporal problems, where subjective time discount rates are typically inferred from choices over when to receive arbitrary rewards, with variations in the structure of the posited discounting function (the range of methods and results have been reviewed by Frederick, Loewenstein, and O'Donoghue 2002; Chabris, Laibson, and Schuldt 2016; Ericson and Laibson 2019; Cohen et al. 2020; Imai, Rutter, and Camerer 2021; Matousek, Havranek, and Irsova 2021). However, the present chapter is interested in intertemporal trade-offs in the specific context of the labor market, in which the relevant choice refers to a recurring payment rule and the payoff is the counterpart of a labor service.

Within the literature on the timing of labor earnings, my findings contrast with the series of studies that manipulate the payment rule for farmers and informal workers in Kenya and Malawi (Brune and Kerwin 2019; Casaburi and Macchiavello 2019; Kramer and Kunst 2020; Brune, Chyn, and Kerwin 2021). Those experiments consistently find that workers prefer a single deferred payment over more frequent, smaller installments. In such a design, however, the choice for later payment is also a choice for a bulky payment, which explains the interpretation that the results reflect a demand for safe savings devices that allow the workers to purchase large indivisible goods. In the present chapter, the contracts differ in the *interval between the work task and the respective pay* (either t + 0 or t + 30). Since neither option allows the accumulation of earnings over multiple days, the results are uncontaminated by potential preferences for lump-sum amounts.

Thirdly, this chapter extends the adoption of quantitative analysis of free text in applied economic research, illustrating how this non-standard data can offer original insights and provide concrete interpretations for conceptual parameters. From a methodological perspective, the present application is closest to the discussion presented by Ferrario and Stantcheva (2022), who use word clouds and keyword analysis to study partisan differences in people's concerns regarding taxation in the United States. For an overview of other recent developments in the analysis of text in economics, see Gentzkow, Kelly, and Taddy (2019) and Ash and Hansen (2023).

Finally, my results complement the ongoing debate on the costs and benefits of platform work, one major case among the increasing menu of alternative work arrangements, as reviewed in Mas and Pallais (2020). While the literature points to flexibility as the primary benefit of the modern gig economy (see for instance Hall and Krueger 2018; M. K. Chen et al. 2019; Oyer 2020; The World Bank 2023; Callil and Picanço 2023), my chapter argues it is also a way to secure income faster, which is a precious feature if workers need (or expect they might need) to address short-term shocks. In this sense, my results are aligned with the findings from Koustas (2018, 2019), who documents that drivers in the United States tend to take up this activity following a period of falling income, decreasing assets, and increasing debt, on average. The rideshare earnings offset part of the lost income, but not all of it, analogous to a safety net.

The remainder of this chapter is organized as follows. Section 2 describes

the operation of the ridesharing activities in Brazil, focusing on the rules that determine the drivers' payout. Section 3 describes the survey design, the preference elicitation method, and the experimental manipulation, and provides an overview of the sample. Section 4 reports descriptive results from my survey, including a profile of the ridesharing drivers and their work routine. The same section also presents a text analysis of the qualitative responses from the drivers. Section 5 reports the experimental results, investigates heterogeneity in the effects for those who drive as a primary or a secondary occupation, discusses the evidence on a potential mechanism, and performs robustness checks. Section 7 concludes with a discussion of the implications of the results and directions for further research.

2. Context

There were at least 1.3 million people actively working as ridesharing drivers in Brazil in the third quarter of 2022, according to the administrative records from the leading platforms (Callil and Picanço 2023). While this group remains a small slice of the total working population (99.3 million), it already represents about 1/4 of the contingent employed by the sectors of accommodation and food services nationwide (5.3 million), or 1/6 of the workers in the construction sector (7.4 million), as per the estimates from the national household survey for the same period (Instituto Brasileiro de Geografia e Estatística 2023).

In essence, ridesharing platforms are companies that use digital applications to intermediate the supply and demand of personal transportation services. When a client requests a ride on such platforms, the task is proposed to available drivers in that geographic area, who can accept it under the posted rates. In this chapter, we define ridesharing drivers as those who supply labor in the form of transportation services under this arrangement.

A crucial attribute of this job is a relatively low entry barrier. To join the pool of active drivers for the major ridesharing platforms in Brazil, one must have a smartphone, no criminal record, and a professional driver's license (which requires psychological tests conducted by the local transit authority). Even though most drivers use their own car to work, this is not a requirement indeed, about 1 out of 4 rent their working vehicles, as I document in the next section. Renting is also an alternative adopted by drivers whose car does not comply with city-level standards for vehicles used in professional transportation.

At the time of the experiment, ridesharing workers in Brazil were in a gray area between regular employees and autonomous service providers from the perspective of labor regulation and social security coverage. They could access the public health system and were eligible for means-tested cash transfers and disability benefits, which are universal welfare policies. However, the social security system only grants contributing workers labor protection benefits (such as temporary work incapacity, maternity leave, and retirement pension). While any platform driver could pay social security contributions as individual own-account workers, this participation was not enforced, and coverage was effectively dependent on the driver's initiative (Center for Education and Research in Innovation 2021). Furthermore, drivers are not subject to the national minimum wage nor work hours restrictions that apply to employees.

From the driver's perspective, rides are priced based on a starting fare, a rate by minute, and a rate by kilometer, subject to a minimum total amount. The exact reference value for each component is specific to the region where the driver operates, as the companies adopt different remuneration rates according to local market conditions. The platforms offer temporary multipliers when demand is high to attract more drivers.

Despite this combination of factors, the bulk of the drivers' remuneration is typically determined by the base rate per kilometer (except for unusual circumstances, such as one-block rides). This is relevant for the purposes of this research, as we exploit the fact that the kilometer rate is a salient earnings component.

Importantly for my research design, the platform has extensive autonomy to set (and to change) the details of their compensation policy, including the base rates and the payment timing, in contrast to most work arrangements. At the time of the experiment, compensation was organized as follows: the passenger pays the platform at the end of a ride, the amount due to the drivers is added to their outstanding balance, and the accumulated value is deposited in the drivers' bank account once a week.

While all the major platforms adopted a similar policy on payment timing at the time of the survey, they were not constrained by technical reasons (a sameday deposit would be equally feasible), legal regulations (the payment standards from the labor code did not apply to ridesharing drivers), nor social norms (there was no longstanding tradition nor strong expectations that ridesharing drivers should be paid weekly). In fact, the leading companies have already introduced mechanisms that allow drivers to access their outstanding balance before the weekly deposit date, but these alternatives require the use of a payment card provided by the platform, which can be subject to transaction fees. There is no public information regarding the drivers' adoption of such payment devices.

3. Experimental design

The survey experiment was implemented with one of the leading ridesharing platforms in Brazil. On the afternoon of January 24, 2023, an invitation to participate was distributed to the mobile phones of all drivers registered with this company. A reminder was sent two days later, and the data collection was concluded on the 31st. Within this period, I documented the participation of

14,265 drivers, making it one of the largest surveys with platform workers to date.

The sample includes all individuals who agreed to participate and were assigned to a treatment or a control group. In practice, it means they answered at least the question regarding the subnational region where they usually work, which is the information required to perform the stratified randomization. The sample excludes (a) 35 cases flagged by the survey software as potentially repeated responses by the same individual and (b) 7 observations coming from the two strata with less than 20 observations. Unfortunately, it is not possible to report the precise response rate to the recruitment because the number of drivers registered with the company is confidential information. For a broad reference of magnitude, there were about 1.27 million active drivers in that period (Callil and Picanço 2023). This figure implies we collected data on about 1.1 percent of the universe. Section 4.1 provides further discussion on how the sample compares to the relevant reference groups in terms of demographics.

The design represents a field experiment in the sense that it targets the relevant subject pool in a real-world context, namely ridesharing drivers evaluating ridesharing contract bundles (Harrison and List 2004). The recruiting message was sent via the ridesharing application itself, and participants could participate in the survey while waiting for their next passenger.

However, the survey was conducted outside the ridesharing application, in a third-party software with a distinct visual identity, to emphasize that the company did not do the data collection. To minimize the risk that people would participate strategically, the recruiting message and the consent form stressed upfront that an academic economist was running the survey to study the drivers' routine and their personal experience with this activity.

3.1. Preference elicitation protocol

While the questionnaire covers a rich set of sociodemographic and work-related variables, the key innovation is the elicitation of the workers' preferences for payment timing. The core question reads as follows:

For some drivers, it is important to be paid for their rides as soon as possible. Others prefer a higher value, even if it takes longer for it to be deposited.

If you could choose, which of those options would work best for you?

[] I'd prefer {*base rate b*} per km, always deposited on the same day of the ride.
[] I'd prefer {*multiplier m × b*} per km, always deposited 30 days after the ride.

The bracketed values were calculated dynamically according to the geographical region of the driver, such that the baseline rate b for the same-day option matches the actual kilometer rate that the respondent is familiar with. The 30 days rate is calculated using a multiplier m to the baseline rate b (1.24 in the first question; 1.06 or 1.96 for the second question; and 1.03, 1.12, 1.48 or 2.92 for the third question, as detailed in figure 1). This strategy ensures that the relative monetary differences are the same at each step regardless of the city where the driver works, even though everybody sees values that are realistic within their own market.



FIGURE I – Sequences of possible contract choices and the corresponding rates. The multipliers were set with the objective of balancing precision (having sufficiently narrow intervals, especially at the bottom of the distribution) and coverage (being able to capture preferences all over the potential distribution), with a minimal number of iterations (3 questions). To that end, the simple rule adopted was to double the marginal percent increase over the tree branches: 3, 6, 12, 24, 48, 96, and 192. The 30 days deferral was chosen to mimic the longest interval without payment that is typical for a wage worker in Brazil.

This measurement strategy (also called "titration", "unfolding brackets", "bisection", "double bounds", or "staircase method") has a long tradition in lab and field applications. It is internally consistent by design and requires only a brief sequence of pairwise choices, which are desirable properties for a mobilebased survey. In essence, the design identifies a range containing the individual indifference point by interactively increasing the value of the option that was not selected before. If the respondent chooses same-day payment, the follow-up question will propose a higher multiplier to the late remuneration; conversely, if they select the late payment, the follow-up question will show a smaller multiplier for this option. Since indifference was not an option, individuals were forced to devote sufficient attention to picking their preferred choice. The unfolding protocol is repeated three times, leading to eight indifference intervals.

The interpretation proposed in this chapter is that each choice provides boundaries for how much the individual values the fast payment option in terms of forgone earnings. In concrete terms, if I take the same-day contract in the first question, I am willing to forgo at least 0.24 out of every 1.24 of my potential earnings per kilometer to have the benefit of being paid faster. Equivalently, this choice implies a lower bound of about 19% for the willingness to pay for this feature — or, more precisely, the *compensated* WTP, as the discussion is about the pay rate per unit of effort, abstracting from possible changes in working intensity. Throughout this chapter, all mentions of WTP should be understood in these terms.

An alternative interpretation would be to frame the results in terms of pure time preference and to infer a subjective monthly discount rate of at least 24% from the aforementioned choice. This chapter favors the use of a WTP framework instead. First, the WTP is agnostic on the underlying links between utility and different choices over deferred payment, while a pure time preference framework requires some extra functional form assumptions. Second, WTP has a natural scale that goes from zero (not willing to renounce any earnings) to almost one (willing to forgo nearly all earnings), while discounting would range from o to positive infinity, imposing additional difficulties on the interpretation of the highest interval in the elicitation scale. More importantly, WTP is a more general concept than pure time preference in that it need not assume that the observed behavior is a consequence of a taste parameter. While heterogeneity in pure time discounting is likely to be a reason behind the choices I document, they are not required to be the only channel, and the measurement choice makes this point more transparent. Finally, reporting the results in terms of WTP puts them on the same scale usually adopted by other choice experiments manipulating job attributes. In practice, if one still favors the time discount perspective, the qualitative results would remain valid, but the magnitudes would require the appropriate conversion following the ancillary assumptions, for instance, using an exponential functional form and a monthly frequency.

One could worry that loss aversion would contaminate the results if the alternatives present values nominally inferior to the ongoing rates, as workers tend to respond strongly against the perception of earnings cuts. To avoid this concern, the choice structure always uses multipliers (of at least 1.03) on top of real-world rates.

Another concern is potential status quo bias if the alternatives include the current payment rule. This risk is not present in this design because the respondent is always choosing between two competing net gains relative to the status quo: either you have your usual rate b, but paid sooner than weekly, or you can have a nominal increase over b, but deferred for a longer time than the current rule.

Finally, note that the choices are designed to avoid, in all scenarios, the possibility of earnings accumulation over multiple working days. This is meant to block the possibility of payments in large chunks, which could confound the results since deferred lump sums are known to be valuable for workers as a commitment device and as savings instruments in themselves (Brune and Kerwin 2019; Casaburi and Macchiavello 2019; Brune, Chyn, and Kerwin 2021). In my design, the interest is solely on the time interval between work and payment; therefore, it is essential to eliminate the accumulation channel.

This chapter acknowledges that reported choices for hypothetical scenarios have limitations. To be clear, respondents received no remuneration to participate in the survey and were informed that their answers would not affect their contracts with the platforms. The critical question is whether voluntary, unincentivized participation could compromise the results. In a methodological discussion, Read (2005) stresses that incentives are not unconditionally necessary nor sufficient for valid results and notes that applied researchers should instead ponder what role a monetary payoff would play in a given elicitation design. In the present case, to recover unbiased results, we require that the subjects understand the alternatives, correctly anticipate their choice, and do not misrepresent their preferences. These assumptions are plausible in this setting because the experiment is close to the subjects' familiar working routine. In other words, I assume that adult drivers do not require extra incentives to understand how kilometers translate into income, can anticipate what consequences a change in the payment timing would have for their household budget, and do not have a systematic reason to distort their choices.

3.2. Experimental manipulation

To measure how the salience of one's financial conditions may affect one's preferences for rapid payment timing, the implementation of the survey splits the respondents into three groups, as shown in figure 2.



FIGURE 2 – Sequence of the survey blocks according to the assignment groups. The randomization was stratified by geographical region, with the regions defined as (a) the capital of the State and the surrounding cities or (b) the remaining cities in the State, for each State in the country.

A third of the respondents are taken as the reference group, in which case people are asked about their sociodemographic characteristics and then invited to choose their preferred contract, following the protocol described above. In treatment group A, respondents are exposed to an additional question block inviting them to discuss how they would deal with an unexpected expense in the amount of R\$ 1,400 (or about US\$ 560 under purchase power parity, slightly above the monthly minimum wage for a full-time job in Brazil). In treatment group B, they are asked how they would spend an unexpected gain of the same magnitude. In both cases, the extra questions take place just before the contract choice.

The objective is to exogenously induce people to an introspective exercise that retrieves the information necessary to react to the problem at hand. Treated individuals do not receive any new data, they are primed to become particularly aware of their circumstances. The critical assumption is that, after the exercise, the financial context examined by the respondent remains readily available in their minds. In this context, two complementary problems (coping with unexpected expenses versus using unexpected income) were designed to pin down which part of the induced salience can explain any systematic difference observed in the reported choices. The treatment consists of the *unexpected expense* scenario:

Imagine you received news of a domestic emergency (an urgent home repair, or a health treatment that cannot wait). Because of this, you will have to disburse R\$ 1,400 more than expected this week.

What is the first word that comes to your mind?

In practice, how would you cover this unexpected expense of R\$ 1,400 right now?

Or the following *unexpected income* scenario:

Imagine you received news of a surprise payment (the result of a lottery or an unexpected refund, for example). Because of this, you will receive an extra deposit of R\$ 1,400 this week.

What is the first word that comes to your mind?

In practice, what would you do with this unexpected income of R\$ 1,400 right now?

Since typing demands more attention and cognitive effort than just clicking or swiping through questions, we can be confident that respondents were engaging with the problems, as also suggested by the time spent in the treatment module. Of all participants actively answering the questionnaire just before the treatment block, 96% typed at least a word in their responses (94% in the expenses arm, 98% in the income arm). Most participants took between 20 seconds and one minute to describe what they would do in the proposed scenario, with a median of 29 seconds in the case of an unexpected expense, and 35 seconds if they had to decide how to spend the surprise income. In both treatment arms, under 2% of the active respondents took less than 30 seconds to go through the whole treatment protocol (that is, vignette, the first word that comes to mind, and what would you do).

Another benefit of applying this treatment to a sample of ridesharing drivers is that they are familiar with smartphones, contributing to the very high compliance. Recurring spelling mistakes, systematic punctuation use, and the occasional emoticon in the responses also reflect a high level of engagement and minimize concerns with computer-generated responses.

4. Descriptive results

This section covers two complementary sets of descriptive results. First, I provide an overview of the sociodemographic characteristics of the ridesharing drivers in the sample, emphasizing that they are similar to the general working population in many dimensions. The sample description also discusses their work routine, their earnings, and the differences between those who drive as a primary or a secondary job.

Next, I characterize the distribution of WTP for same-day earnings among the participants, as measured in the main elicitation protocol. Two findings stand out: there is a wide dispersion of preferences, with at least 5 percent of workers in each possible WTP interval that we observe, but they are strongly overrepresented at higher buckets, with WTP of 32% or more. The analysis of associations between the WTP and other attributes, in particular their total household income per capita, supports the interpretation that such distribution is partially driven by structural material scarcity.

4.1. Who are the ridesharing drivers?

The ridesharing drivers in this study are predominantly young adults (52.4% are less than 38 years old), identify themselves as black or mixed-race (62.8%), and have no college degree (83.9%). In most cases, they live with another adult (57.6%) and at most one child (70.3%). Even though the subjects were sampled from the active drivers of a particular company, they are similar to the general population of platform drivers in the country, according to the observable demographics reported by other sources. The reference closest to the present study is Callil and Picanço (2023), who conducted telephone interviews with 1,518 drivers between August and November 2022. They report that 56% of them are under 39 years old, 62% are black or mixed race, and 81% have no college degree. They do not report statistics on household composition.³

What is more remarkable, the drivers in this study also reflect the general diversity of the urban, adult, working population in Brazil in terms of ethnicity, age, education, and household composition, as documented in table 1. While it would be excessive to claim that the participants of this study are representative workers in a broad sense, this tabulation shows that they are not a particularly eccentric group.

^{3.} Other studies have used the combination of category ("own-account worker"), reported occupation ("driver"), and activity ("passenger transportation") to identify platform drivers in the National Household Survey (PNADC), notably Góes, Firmino, and Martins (2021) and Góes, Firmino, and Martins (2022). While useful as an initial approach, it is not the ideal proxy because it does distinguish platform drivers from taxi drivers. For this reason, I still use PNADC to characterize the adult urban workforce in Brazil, for which it remains the best source in this context, but I avoid using it for the purpose of studying the platform drivers in particular.

		 Ridesharing Drivers Survey						National Household Survey (PNADC)					
	All drivers		Driver as main job		Driver as secondary job		Adult urban workforce		Adult urban own-account workers		Adult urban employees		
	stat.	s. e.	stat.	s. e.	stat.	s. e.	stat.	s. e.	stat.	s. e.	stat.	s. e.	
Gender (share in %)													
Male	93.2	(0.2I)	92.7	(0.30)	93.9	(0.46)	54.8	(0.14)	63.2	(o.35)	52.6	(0.20)	
Ethinicity (share in %)													
Black	13.4	(0.29)	13.1	(o.39)	I4.0	(o.67)	11.3	(0.16)	10.7	(0.29)	11.8	(0.20)	
Mixed-race	49.4	(0.42)	49.0	(0.57)	47.9	(0.96)	43.I	(0.27)	43.2	(0.45)	42.7	(0.31)	
White	37.3	(0.4I)	37.9	(0.55)	38.1	(0.94)	45.6	(0.30)	46.1	(0.49)	45.6	(o.33)	
Age group (share in %)													
18 to 27 years old	I 4. I	(0.30)	15.0	(0.4I)	I2.I	(o.63)	23.I	(0.18)	I4.4	(0.3I)	24.2	(0.23)	
28 to 37 years old	38.3	(0.4I)	39.1	(0.55)	37.I	(0.93)	26.6	(0.2I)	25.4	(o.39)	27.8	(0.26)	
38 to 47 years old	31.5	(o.39)	29.9	(0.52)	35.I	(0.92)	24.5	(0.18)	24.9	(o.35)	25.I	(0.22)	
48 to 57 years old	12.2	(0.28)	12.0	(o.37)	12.0	(o.63)	16.9	(0.15)	20.0	(0.30)	16.2	(0.18)	
58 years old or more	4.0	(o.17)	4.0	(0.22)	3.7	(0.36)	8.9	(0.12)	15.2	(0.29)	6.7	(0.12)	
Education (share in %)													
Primary education or less	II.I	(0.27)	10.9	(o.35)	8.3	(o.53)	24.I	(0.23)	32.7	(0.4I)	21.0	(0.25)	
Some high school	7.9	(0.23)	8.2	(0.31)	5.7	(o.45)	6.7	(o.11)	7.I	(0.2I)	6.2	(0.12)	
High school	44.I	(0.42)	44.7	(0.57)	43.I	(0.95)	38.1	(0.24)	36.2	(o.39)	38.2	(0.29)	
Some college	20.7	(o.35)	21.4	(o.47)	20.5	(o.78)	7.3	(o.11)	5.3	(0.18)	8.0	(o.14)	
College or above	16.2	(0.32)	14.8	(0.40)	22.5	(o.8o)	23.8	(0.3I)	18.7	(o.43)	26.7	(0.35)	

TABLE 1 – Characteristics of the drivers in the survey and corresponding summaries for urban adult workers in Brazil

	Ridesharing Drivers Survey						National Household Survey (PNADC)					
	All drivers		Driver as main job		Driver as secondary job		Adult urban workforce		Adult urban own-account workers		Adult urban employees	
	stat.	s. e.	stat.	s. e.	stat.	s. e.	stat.	s. e.	stat.	s. e.	stat.	s. e.
Household composition												
N. of adults (age 18+)	2.4	(o.oı)	2.4	(o.oı)	2.4	(0.02)	2.5	(0.01)	2.4	(o.oı)	2.5	(o.oı)
N. of kids (age < 18)	I.0	(0.01)	I.0	(0.01)	I.0	(0.02)	o.8	(0.01)	0.8	(0.01)	0.8	(o.oı)
Work routine												
Work hours per week	53.0	(0.24)	60.1	(0.26)	32.9	(0.39)	39.7	(0.05)	38.0	(0.13)	40.0	(0.05)
Monthly income (in R\$)												
Average work income	2,267	(15)	2,501	(17)	1,597	(23)	2,805	(28)	2,293	(32)	2,743	(28)
Average household inc. per capita	1,381	(12)	1,333	(13)	1,517	(25)	2,084	(23)	1,987	(28)	2,143	(25)
How long in this job (share in %)												
Less than 3 months	12.2	(0.31)	10.3	(o.35)	16.6	(0.72)	10.9	(0.14)	8.6	(0.24)	12.3	(o.17)
3 to 6 months	10.0	(0.28)	9.3	(o.33)	I2.2	(0.63)	6.5	(o.11)	4.6	(0.22)	7.5	(0.13)
6 months to 1 year	11.7	(0.30)	11.7	(o.37)	I2.I	(o.63)	6.3	(o.11)	4.I	(o.17)	7.4	(o.14)
I to 2 years	16.8	(o.35)	16.1	(0.42)	18.1	(0.74)	10.8	(0.14)	7.9	(0.23)	12.3	(o.17)
2 to 4 years	29.4	(0.42)	30.5	(0.52)	26.4	(o.85)	22.2	(0.17)	23.1	(o.34)	22.I	(0.20)
More than 4 years	19.8	(0.37)	22. I	(o.47)	14.7	(o.68)	43.4	(0.23)	51.7	(0.42)	38.5	(0.26)
Social indicators (share in %)												
Contributes to a pension system	43.0	(o.53)	31.2	(o.58)	76.1	(0.91)	67.4	(0.23)	33.5	(o.44)	79.8	(0.23)
Household inc. per cap. < USD 5.5/day	11.3	(0.32)	12.2	(o.39)	8.4	(o.56)	8.5	(0.15)	8.4	(0.23)	4.8	(o.11)

TABLE I – Characteristics of the drivers in the survey and corresponding summaries for urban adult workers in Brazil (continued)

		Ridesharing Drivers Survey					National Household Survey (PNADC)					
	All drivers		Driver as main job		Driver as secondary job		Adult urban workforce		Adult urban own-account workers		Adult urban employees	
	stat.	s. e.	stat.	s. e.	stat.	s. e.	stat.	s. e.	stat.	s. e.	stat.	s. e.
Country region (share in %)												
North	8.8	(0.24)	8.5	(0.32)	8.3	(o.53)	7.5	(0.13)	8.7	(0.22)	6.9	(o.14)
Northeast	20.0	(o.34)	20.3	(0.46)	19.3	(0.76)	21.4	(0.24)	23.I	(o.38)	19.7	(o.27)
Southeast	46.7	(0.42)	48.0	(0.57)	44.6	(0.96)	47.7	(o.34)	46.I	(0.51)	48.7	(0.40)
South	13.6	(0.29)	12.8	(o.38)	16.4	(0.7I)	14.8	(0.20)	I4.I	(0.29)	15.6	(0.24)
Central-West	10.9	(0.26)	10.4	(0.35)	II.4	(o.61)	8.6	(o.14)	7.9	(0.19)	9.1	(o.17)
Survey sample												
Number of observations	Ι4,	265	7:	74I	2,	708	133	,762	31,	270	83,	369

TABLE I – Characteristics of the drivers in the survey and corresponding summaries for urban adult workers in Brazil (continued)

Notes: [1] The drivers' survey was conducted by the author between the 24th and the 31st of January 2023 and its underlying population is all drivers working with a leading ridesharing company in Brazil. [2] The figures regarding to the general workforce are calculated using the microdata from Brazil's official labor survey, refer to the full year of 2022, and are weighted to be representative of the active population above 18 years old and living in urban areas. In particular, I use the data collected by PNADC's 5th interview with the sampled households, which records household income from all sources. [3] For all variables and all subpopulations, the statistics are calculated using the available responses required for that specific item, and therefore the number of observations may vary for different attributes. The sample size for all drivers represents to the number of unique individuals who participated in the survey, while the combined number primary job drivers and secondary job drivers refer to the respondents for whom there is sufficient information for this breakdown. [4] Monetary values from PNADC are reported in January 2023 equivalent terms. [5] Work-related statistics (such as *work income, work hours* and *how long in this job*) are specific to the occupation indicated in the column. [6] The *household income per capita* is composed of all income sources from all invidividuals in a given household. [7] Non-male drivers are composed by 6.7% of female drivers and o.1% of respondents who do not identify neither as male nor female; PNADC has no comparable gender information.

The striking exception is that males represent 93.2% among the ridesharing drivers, in contrast to 54.8% in the urban workforce. However, such gender unbalance is typical for this industry, particularly in low- and middle-income countries. The International Labour Office reports that females make up, on average, 5% of the ridesharing drivers in Chile, Ghana, India, Indonesia, Kenya, Lebanon, Mexico, Morocco, and Ukraine (International Labour Office 2021). Looking at the base of Uber drivers in the United States, Cook et al. (2021) document a female share of 27.3%, with the caveat that the proportion of active female drivers at any given month is lower than that because women leave the job at a higher rate (76.5% of them are no longer active within six months, compared to 65.0% for men).

For completeness, table 8 in the appendix replicates the descriptive statistics from table 1 but keeps only men in both the drivers' sample and in the general workforce. In this case, after removing the women from the comparison, the share of workers with college decreases, while the average work hours and work income increase for all subgroups.

The drivers report an average net income from ridesharing of R\$ 2,267 per month, after regular working expenses, which is equivalent to about US\$ 900, adjusting for purchase power parity (see figure 3 for the distribution of monthly earnings from ridesharing). This average value represents 1.7 times the national minimum wage for a full-time formal employment position in Brazil. On the other hand, it is about 20% below the average monthly earnings reported by the general workforce in the same period (or 26% less, if we compare only male drivers with the male working population), as measured by the national household survey.



FIGURE 3 – Distribution of monthly earnings for the ridesharing activity and the average work earnings for selected reference groups. The dashed lines mark the average work earnings for the different reference groups. The underlying values can be found at table 8 in the appendix.

Going beyond the general average, it is possible to identify two very distinct profiles in this population: 3/4 of the drivers engage in ridesharing as their sole or main occupation (in the sense that it represents their main income source), while 1/4 use it as a supplementary activity. Primary job drivers report working an average of 6 days per week and 10 hours per day, with net earnings of R\$ 2,500 per month (US\$ 1,000 PPP). In contrast, secondary job drivers drive 4.4 days per week and 7.2 hours per day, with net earnings of nearly R\$ 1,600 per month (US\$ 640 PPP).

These figures imply that secondary job drivers are able to earn about 14% more per hour (US\$ 4.2 vs. US\$ 4.9 in PPP terms), suggesting that they are able to optimize their driving routine, choose more profitable periods, or to respond more strongly to changes in demand compared to main job drivers, who work more regularly.

My data does not allow us to conclude if Brazilian ridesharing drivers follow some form of earnings targets, as proposed in the lively literature on the labor supply of taxi drivers (Camerer et al. 1997; Farber 2008; Crawford and Meng 2011; Thakral and Tô 2021). However, the behavior of primary job drivers appears to be consistent with a maximization of their total monthly earnings, instead of their hourly gains. Since most drivers in this group tend to work more than 8 hours per shift, we can conclude that they regularly find the marginal revenue from the 9th hour more valuable than going back home in a typical working day.

The polarization between those two types of drivers is also reflected in other dimensions, as primary job drivers are systematically younger, less educated, live in poorer households, and are less likely to contribute to a pension system. Yet, these two groups have a major feature in common: *both appreciate the fact that this activity offered them a way to secure some income quickly*. Indeed, this is the single most frequent reason mentioned by the respondents when asked about what motivated them to take up ridesharing, considering the other paid activities they could do, as detailed in figure 4.

This is an important result because it complements the usual argument that points to flexibility and autonomy as the major differential benefits from the ridesharing activity (see Hall and Krueger 2018; Oyer 2020; The World Bank 2023; Callil and Picanço 2023). It is unclear how the order of importance reported in similar surveys would be affected if they had included an explicit option about quick payments.





(b) Secondary job drivers

FIGURE 4 – Most important reasons for taking up ridesharing, by driver type. The questionnaire presented this set of alternatives in random order to the respondents to avoid sequence bias. The total share of responses may add to more than 100 percent, as people could choose more than one option.

The caveat about these results is that the wording "a way to secure income quickly" can potentially cover two distinct senses for "quick": (a) the low entry barrier that allows people to start working faster relative to the counterfactual of searching for a match with a company and (b) the short time between the work and the associated payment. Both are likely to be present, as discussed in Scarelli and Margolis (2023), but the distinction between them is substantive.

In the next section, we take this investigation a step further with the results from the WTP protocol, which have the double benefit of eliminating ambiguity (by isolating the value of the payment timing only) while being more precise regarding its importance (by measuring it in terms of forgone earnings).

4.2. How much do drivers value same-day remuneration?

The main finding from the preference elicitation protocol is that the possibility of quickly converting labor into cash is extremely valuable for ridesharing drivers. The median driver would rather be paid the same day than 1.48 times as much in 30 days, a choice that suggests a WTP of at least 32% (from foregoing 48 cents out of each 1.48 monetary units of payment). Almost 1 in every 4 drivers would take the same day pay against nearly 3 times as much for the 30-day delay, with an implied WTP of at least 66%. Taking the midpoint of each interval weighted by their mass, the mean WTP is close to 40%.



FIGURE 5 – Distribution of preferences for same-day remuneration

High inflation and high interest rates could be trivial reasons motivating people to avoid deferred payments. However, we can reject that these concerns rationalize the bulk of the behavior documented here, given the magnitude of the multipliers proposed for future payments. For reference, at the time of the data collection, headline inflation in Brazil was under 0.4 percent per month, and food inflation was under 1 percent per month (Ferreira et al. 2023). Similarly, the baseline interest rate in the financial system was around 1 percent per month. All in all, these reference rates mean that the present value of the later payment option in real terms should be adjusted by no more than a few percentage points and thus cannot explain any choice beyond the very first bucket at the bottom of the distribution.

Instead, these extreme preferences appear to partly reflect a context of structural resource scarcity and missing financial instruments, which makes one's labor a source of both domestic solvency *and* liquidity. This view is supported by the monotonic association between contract choices and poverty: the lower the total household income per capita, the more valuable the option to access one's earnings the same day, as summarized in figure 6.

A similar correlation emerges at the geographical level. There is a known gradient in the median regional income (and in other poverty indicators derived from the surveys of the national statistics office) going from the Northern (poorest) to the Southern (richest) regions of the country. Since I collect data from drivers in all regions, I can document that a similar gradient holds for the WTP for same-day remuneration but in the opposite direction, as depicted in figure 7. Under the mild assumption that the potential association between the outcome and the response rates is not region-specific, the results suggest that drivers in the poorest regions are the ones who favor quick payment the most.



FIGURE 6 – Distribution of preferences for same-day remuneration by quintile of household income per capita



(a) Average WTP, as measured in the drivers survey

(b) Median household income per capita, as measured in the PNADc



Compared to the valuation of other job amenities documented in the literature, the amount people are willing to give up for same-day remuneration is indeed at the high end, but it is not implausible. Manipulating the application process for a position in a call center in the United States, Mas and Pallais (2017) find that the applicants were willing to forgo 20% of their wages to avoid a schedule set by an employer on short notice, and 8% for the option to work from home. Using a panel of Danish respondents, Eriksson and Kristensen (2014) estimate a 13% WTP for high job flexibility, 8% for 5 days of training, and 7% for a large health package. With a sample of undergraduate students from the New York University (NYU), Wiswall and Zafar (2018) document that female students report a WTP of 4% for a percentage point lower chance of being fired, and a WTP of 7% for the option of working part-time. Looking at how much the workers in India are averse to accepting tasks that do not align with their own identity, Oh (2023) finds that 43% are willing to forgo at least 10 times their daily wage to avoid a type of work that is associated with other castes.

Part of the dislike of being paid later may also come from a concern that the company might fail to honor its payment commitments at some point, willingly or not. This factor is unlikely to play a large role in the results because most drivers already have a track record of many months working with this firm, which contributes to building trust in its payment capacity. As suggestive evidence of this, I find that those who have just started the activity tend to have the highest WTP, as depicted in figure 9. However, this result should be taken with a grain of salt, given that newcomers could plausibly be the most liquidityconstrained. In any case, there is little systematic association between seniority and WTP beyond the very first month of experience. Furthermore, according to Brazilian commercial law, the workers, service providers, and contractors have priority in the event of a business liquidation. Hence, the risk of non-payment due to the company quitting the market should not explain the magnitude of results I find, and much less so for a short 30-day interval.

There could also be some habit formation regarding payment rules, in the sense that people tend to lean towards the schemes they are already familiar with. If this is true in this setting, drivers who were previously wage employees (and, thus, subject to a form of deferred payment) could show less resistance to the 30-day interval relative to drivers who had other occupations. Yet, if we break down the preferred contract choices by the previous state, we find little difference between those coming wage-employment or self-employment. Instead, the major contrast in this dimension comes from those with or without any form of paid activity, with the inactive and the unemployed showing the highest willingness to pay for same-day remuneration, as per figure 9.



FIGURE 8 – Preferences for same-day remuneration by demographics



FIGURE 9 - Preferences for same-day remuneration by work profile

Having reviewed some alternative explanations for our findings, I argue that these results reflect primarily a condition of liquidity constraints from the workers' perspective. What is more, even if the choices reported by the Brazilian drivers imply a very high urgency for liquidity, they should not be taken as an anomaly from an exotic setting. To see this point, we may refer to the use of payday loans in the United States, a practice that shares many economic similarities with the patterns we find here despite taking place in a different context. These are short-term advances on a worker's future paycheck, typically over small amounts (80% of them being under US\$ 300), with implicit interest rates between 400 and 1,000 percent per year (Stegman 2007), which would be equivalent to 14% to 22% in compounded monthly rates. The conditions for such loans are so extreme that 16 jurisdictions prohibit payday lending under variations of usury laws, but even formal limitations have only a modest effect on actual access to it, as 12% of the consumers in these jurisdictions have taken payday loans at least once in the past 5 years, often through online lenders (Harvey, Robb, and Peterson 2024). In total, 12 million American adults use payday loans annually, most frequently to cover recurring expenses, such as utilities, credit card bills, rent or mortgage payments, or food (the main reason for 69% of the first payday loans), followed by unexpected expenses, such as a car repair or emergency medical expense (16%) (The Pew Charitable Trusts 2012).

Also in the US context, McDevitt and Sojourner (2023) show that people are willing to pay high fees to access the funds from their paper checks faster. Given a choice between depositing a check (and waiting for it to clear through the banking system) and cashing it (for a fee), an extra day of check-clearing time makes the average account holder 65% more likely to cash it. They estimate that the average customer is willing to pay the equivalent of US\$ 11.17 per day to get their cash faster, which implies an annualized discount rate of 11,054% for the average check, or 48% per month. In line with our results, they find that such willingness to pay for liquidity is higher among households with the lowest income.

Finally, we can ask what consequences such a strong preference for liquidity could have for the platforms that intermediate the services performed by these drivers. The magnitude of the results allows us to conclude that the companies are most likely saving money by offering payment intervals well under a month (that is, daily or weekly, depending on whether the drivers use the company-issued payment card). To see that, let us perform a simple counterfactual exercise taking at face value the preferences reported by the drivers. If a company were to implement the 30-day rule and pay each driver using the rate they would demand to make this change, the nominal payroll would increase by about 67%,

assuming no differential responses in labor supply.⁴

While useful for illustrating the magnitude of these preferences in monetary terms, this is admittedly an extreme scenario. Some drivers at the high end of the WTP distribution would take the deferred payment at rates lower than their preferred ones if that was the only option on the table, a scenario that the elicitation does not cover. However, a large share of them would simply not drive if a short payment interval were not available. The best proxy for this group is the 24% who never took the deferred payment in the experiment. In any case, fast payment schemes seem to allow the platforms to attract more drivers than they would otherwise, which is a strategic need of this business given that a large pool of drivers leads to a lower waiting time for the passengers. This is possible because the returns the platforms would have by keeping the cash for longer (say, via traditional financial instruments) are much lower than the value the workers assign to getting it sooner.

5. Experimental results

This section presents the findings related to the experimental manipulation module. It starts by defining the working sample adopted in the different treatment effect estimations and discusses the randomization balance over the treatment arms.

After that, I apply text analysis to the open-ended responses provided by the drivers as part of their treatments. Since these techniques are not yet standard tools in Economics, I briefly discuss the decisions involved in the process of text cleaning before reporting the patterns of liquidity constraints that emerge from the keywords used by the individuals with the strongest payment urgency.

The core of this section is dedicated to analyzing the treatment effects. The main results suggest that both treatments (either a discussion about emergency expenses or the use of unexpected income) induce ridesharing drivers to decrease the importance they assign to immediate payment and increase the probability they choose a larger payment instead. The section concludes with a discussion of potential cognitive mechanisms behind such an effect.

5.1. Working sample and treatment randomization balance

Given the nature of the data collection, it is reasonable to expect a gradual attrition throughout the questionnaire. The drivers may receive an offer to pick up someone or may want to check an incoming message on their mobile phone, among many other reasons, leading them to drop out at some point. With that

^{4.} This figure can be calculated using microdata at the individual level and is linked to the average willingness-to-pay of 40% reported above by the WTP definition: 0.4 = (1.67 - 1)/1.67.

in mind, the survey was designed to be concise, and achieved a relatively high completion rate. Of the 14,265 individuals who responded to the first question, about two-thirds finished it.

From the perspective of the treatment effect estimation, the main concern is that attrition affects the randomization balance between the different arms. As the first step to address this issue, figure 10 plots the number of respondents by treatment group throughout the survey. Participation is consistently very high in all arms over the initial demographic question. However, respondents were slightly more likely to quit after being asked about how they would address a financial emergency (the red line in the plot), while those facing a hypothetical scenario with a surprise income were less likely to drop out (in green), in comparison to the control group (in blue).



FIGURE 10 – Number of active respondents throughout the survey, by treatment condition. A respondent is considered to be active until the last question they answer. Dotted lines are used to signal a question that was not part of the questionnaire in that particular treatment arm.

To investigate if the differential response rates are affecting the sample composition, we look at the characteristics of the respondents. The statistical summaries presented in table 2 suggest that, while people in the expense treatment condition were marginally more likely to drop out, this attrition was not driven by a particular profile of respondents. Formally, we can reject that the set of attributes we observe are jointly significant to distinguish those who completed the survey within this treatment arm. However, that is not the case for the income treatment. The excess of responses recorded in this group is particularly linked with full-time drivers (and, by extension, those who were previously unemployed, work more hours, and do not contribute to social security).

	Control group (n = 2,672)	Treatmen unexpect (n =	nt group A: ed expense 2,597)	Treatment group B: unexpected income (n = 2,873)		
	mean (I)	mean (2)	p-value (1) = (2)	mean (3)	p-value (1) = (3)	
Gender and ethnicity						
Male	0.94	0.92	0.053	0.93	0.324	
Ethinicity						
Black	0.12	0.12	0.836	0.14	0.253	
Mixed-race	0.49	0.49	•	0.48		
White	0.39	0.38		0.39		
Age group						
18 to 27 years old	0.16	0.14	0.327	0.16	0.976	
28 to 37 years old	0.39	0.40		0.39		
38 to 47 years old	0.31	0.32		0.31		
48 to 57 years old	0.11	0.11		O.II		
58 years old or more	0.03	0.04		0.03		
Education						
Primary education or less	0.09	0.09	0.833	0.09	0.869	
Some high school	0.07	0.07		0.08		
High school	0.45	0.44		0.43		
Some college	0.21	0.22		0.22		
College or above	0.18	0.18		0.17		
Household composition						
N. of adults (age 18+)	2.38	2.40	0.606	2.36	0.366	
N. of kids (age < 18)	1.03	1.04	0.908	1.04	0.818	
Other jobs						
Driver only	0.62	0.62	0.652	0.67	0.001	
Driver and employee	0.20	0.20	•	0.18		
Driver and self-employed	0.18	0.17	•	0.15	•	
Previous status						
Inactive	0.03	0.03	0.067	0.04	0.000	
Unemployed	0.27	0.30		0.33		
Self-employed	0.23	0.23	•	0.21		
Employee	0.38	0.36	•	0.34		
Other status	0.09	0.09		0.09		
Income						
Income from this work	2,283	2,324	0.201	2,239	0.185	
Total household income	4,022	4,096	0.285	3,756	0.001	

TABLE 2 – Summary statistics and randomization balance

	Control group (n = 2,672)	Treatmen unexpect (n =	nt group A: red expense 2,597)	Treatment group B: unexpected income (n = 2,873)		
	mean	mean	p-value	mean	p-value	
	(1)	(2)	(I) = (2)	(3)	(1) = (3)	
Work routine						
Work days per week	5.57	5.60	0.439	5.67	0.020	
Work hours in a working day	9.21	9.07	0.024	9.26	0.428	
How many apps	2.03	2.00	0.178	1.98	0.004	
Vehicle ownership						
Rented from friend, family	0.11	0.12	0.460	0.13	0.256	
Rented from agency	0.12	0.11		0.12		
Own car, still paying	0.57	0.57		0.56		
Own car, fully paid	0.19	0.20		0.19		
How long in this job						
Less than 1 month	0.02	0.03	0.469	0.02	0.543	
1 to 3 months	0.10	0.09	•	0.09	•	
3 to 6 months	0.10	0.10		0.10		
6 months to 1 year	0.12	0.11		0.13		
1 to 2 years	0.16	0.15		0.17		
2 to 4 years	0.30	0.29		0.30		
More than 4 years	0.20	0.22	•	0.20		
Share of work income usually saved						
Less than 10%	0.73	0.69	0.002	0.74	0.376	
Between 10% and 25%	0.18	0.21		0.18		
More than 25%	0.09	0.10	•	0.08		
Social security						
Not currently contributing	0.52	0.52	0.686	0.57	0.002	
Public system (as individual)	0.22	0.23		0 .2 I		
Public system (as employee)	0.16	0.16		0.15		
Private system	0.03	0.02		0.02		
Does not know	0.07	0.07	•	0.05		
Country region						
North	0.08	0.08	0.986	0.08	0.998	
Northeast	0.20	0.20		0.20		
Southeast	0.47	0.47		0.47		
South	0.13	0.13		0.14	•	
Central-West	0.11	0.11		0.11		
Mobile phone						
Android 8 or below	0.03	0.04	0.171	0.04	0.565	
Android 9	0.05	0.05		0.05		

 TABLE 2 – Summary statistics and randomization balance (continued)

	Control group (n = 2,672)	Treatmen unexpect (n =	nt group A: red expense 2,597)	Treatment group B: unexpected income (n = 2,873)		
	 (1)	mean (2)	p-value (1) = (2)	mean (3)	p-value (1) = (3)	
Android 10	0.18	0.17		0.16		
Android 11	0.24	0.23		0.24		
Android 12	0.27	0.28		0.28		
Android 13	0.04	0.04		0.04		
iPhone	0.19	0.19		0.19		
oint significance test						
p-value		0.	.122	0.0	000	

TABLE 2 – Summary statistics and randomization balance (continued)

Notes: [1] The baseline sample is composed of the drivers with valid observations for all attributes displayed in the table. [2] For attributes represented as continuous or binary variables, the p-values refer to the statistical significance test of equality of means between the control group and each of the two treatment groups. It is calculated using an OLS regression of the variable on treatment indicators, with standard errors clustered at the sub-state geographical level, according to the experimental design stratification. [3] For attributes measured as factor variables, the p-value is calculated using a pairwise chi-squared test of independence between the control group and each of the two treatment groups. [4] The joint significance test reports the p-value associated with the F-test from a regression of the treatment indicator on all covariates displayed in the table.

Given the slight excess of primary-job drivers in one of the treatment groups, I favor the estimation techniques that use the available information about the drivers to mitigate the consequences of this imbalance. In practice, it means that the working sample needs to be restricted to the 8,142 individuals for whom we observe the full set of covariates that will serve as controls, which are the ones described in table 2.

5.2. Text analysis

This section serves two purposes. On the one hand, it documents how the respondents are reacting to the treatment questions. In this respect, the evidence suggests that the vast majority of the participants invested the necessary effort to provide meaningful answers when primed to do so. Since the differential exposure to this exercise is precisely the dimension manipulated by the experiment, this analysis opens the treatment black box and provides confidence that it is triggering a response.

On the other hand, by leveraging the information recovered through the open-ended questions, it is possible to investigate further the structural reasons behind the dispersion in preferences documented above. While descriptive in nature, the analysis of the words mentioned by the drivers provides a foundation for the analysis of the underlying determinants of the preferences for quick payment.

The quantitative methods adopted here require the transformation of text strings into high-dimensional count vectors (Ash and Hansen 2023). In essence, the idea is to build a matrix where lines represent individual responses and the columns represent the universe of terms that were mentioned in the sample.

In the present case, individual response is defined as the combination of their answers to both questions that make up the treatment (that is, *what is the first word that comes to mind?* and *what would you do?*). In total, 8,507 individuals typed at least one word in their answers, with over 7,000 unique raw words.

The cleaning consists of harmonizing these terms. As a first step, all characters are transformed to lowercase (for example, "App" to "app"), punctuation and diacritical marks are removed ("gratidão" to "gratidao"). Next, I split words that are unintentionally merged ("boahora" to "boa hora"), correct general misspellings ("poblema" to "problema"), and remove stopwords (frequently used ancillary terms that carry little information by themselves, such as demonstrative pronouns). Finally, I keep a single form for words that can be inflected in Portuguese, undoing number declension ("atrasadas" to "atrasada"), gender declension ("atrasada" to "atrasado") and verb conjugation ("adoraria" to "adorar"). The resulting 1,647 terms are translated to English, for presentation purposes, favoring expressions that are closest to the particular context of this survey.

After this cleaning protocol, we recover two distinct matrices, one for each alternative treatment. The first matrix comprises 1,017 unique terms used by 4,157 drivers when describing their reactions to the hypothetical financial emergency. The top 200 terms in this set are summarized in figure 11, in which size and color intensity are proportional to how often the drivers mention them.



FIGURE II – Most frequent terms mentioned by drivers when discussing how they would cover an unexpected expense. The word cloud depicts the 200 most frequent terms used by the ridesharing drivers who were invited to consider a situation where they would need to disburse R\$ 1,400 (US\$ 560 PPP) more than expected that week. The size and color intensity are proportional to the incidence of the term. The same representation using the original terms in Portuguese is available in figure 15 in the appendix.

Two concepts stand out in this graphical representation of total frequency: "work" and "loan". This pattern suggests that (a) precautionary savings are often modest or missing among this population and (b) work intensity is a primary margin of adjustment in reaction to negative liquidity shocks. If that is the case, it means that the possibility to choose your hours and quickly cash them can serve as an insurance mechanism.

Taking their responses seriously, how much extra work do people have in mind? A simple extrapolation using the net earnings from section 4 implies that the average driver would need about 130 working hours to make up the R\$ 1,400 proposed in the scenario (about 2 or 3 working weeks).

Looking at the mirror image of this problem, the second set of answers includes 1,244 unique terms used by 4,350 drivers to discuss what they would do with an unexpected cash windfall. The word cloud shown in figure 12 is dominated by a single term: "pay". In this context, the most common reaction appears to be guided by concerns with recurring household bills and outstanding debts.

celebration extra remedy receive necessity inside exit provisions driver anticipate supermarket really believe invoice course today complete moment return finance debit spend reserves liaht app market because college savings material card tranguility also clothing achieve blessing know error installment food excellent win right evervthina payoff wonderful value return miracle verify nather truth well place enter keep be tami Jesus amen bill improve child me month thing hοι se father first give travel deposit get yield fuel take need pass why thanks lord life still lookfor wife have future jov part aood monev difficult settle hoax surprised son glory maturity more luck fix enough vear until remain hour stop gift power come tax apply pending change icle figure too hallelujah save new dav without great only rati tude lack thiswav help not app g credit daughter certainty little mother happiness leave stay review maintenance small yay use situation arrive alleviate school installment weekly market safe advance always monthly water remaining expense legal run document nothing doubt emergency mistake recognition laughter walk continue health ride something want origin owe passenger unforeseen

FIGURE 12 – Most frequent terms mentioned by drivers when discussing what they would do with an unexpected income. The word cloud depicts the 200 most frequent terms used by the ridesharing drivers who were invited to consider a situation where they would receive an unexpected deposit of R\$ 1,400 (US\$ 560 PPP) that week. The size and color intensity are proportional to the incidence of the term.

It is interesting to note that the religious terms are clearly present in both scenarios. In the first case, "God" comes to mind as a potential source of relief given the financial struggle, while religious terms show up associated with expressions of gratitude in the second group. Likewise, family members are mentioned in both circumstances, as the primary social network available during emergency situations and to share the windfall.

While word clouds are useful for highlighting the predominant topics, they must be complemented with other strategies that are better suited to uncovering the associations between the responses and other observable features. In particular, we want to study which terms are disproportionally adopted by individuals who also show a very strong preference for quick payment rules.

For simplicity, I divide the drivers into two groups: the top third of the distribution (those who claim to prefer same-day payment over 2 or 3 times larger rates) and the rest. The keyword analysis, in this case, is analogous to a chi-square test for a contingency table, in which we study whether a given term is statistically overrepresented in one of the groups. The higher the chi-square statistic, the stronger the evidence against the null hypothesis that a given term is equally likely to be used in both groups. If the term appears in excess among people with high WTP for same-day remuneration, the test statistic is positive (depicted by the red lines in the keyword plots), and it is negative otherwise (the blue lines in the plots).

The results show that the people who rely on family members and on their own labor to help them fix a financial emergency are more likely to prioritize fast payment, as summarized in figure 13. On the other end, drivers who already have credit cards and precautionary funds available are the ones favoring larger earnings.

Similarly, the terms describing potential uses of the unexpected income reflect a strong polarization between circumstances of pressing needs (drivers claiming they would spend their cash windfall procuring food for their house-hold tend to have the strongest preferences for same-day payment) and precautionary behavior (drivers who would save the money for the future also favor contracts with larger, deferred payments), as shown in figure 14.

A critique to this type of analysis is that words lose much of their meaning outside a sentence. While this remains an important caveat in this chapter, the concern is partially mitigated by the constraints imposed by the text collection strategy. We have the benefit of recording responses that are not bounded by a small pool of close-ended alternatives, while being sufficiently tied to the context to give us confidence in their interpretation. For instance, if we had a random sample of twitter posts, it would be hard to interpret the excess of terms like "family", compared to our case where they appear in the reaction to a particular financial scenario.



FIGURE 13 – Keywords from the *liquidity* discussion that distinguish the drivers with the strongest preference for same-day payment. The plot includes terms that were mentioned by more than 0.1% of the individuals and have a chi-squared statistic of at least 3.84, the critical value for 5% significance in a test with two groups. The break in the vertical axis is a reminder that all terms with a statistic in the interval [-3.84, 3.84] are omitted.



FIGURE 14 – Keywords from the *consumption* discussion that distinguish the drivers with the strongest preference for same-day payment. The plot includes terms that were mentioned by more than 0.1% of the individuals and have a chi-squared statistic of at least 3.84, the critical value for 5% significance in a test with two groups. The break in the vertical axis is a reminder that all terms with a statistic in the interval [-3.84, 3.84] are omitted.
5.3. Average treatment effects

This section investigates whether the salience of the workers' financial circumstances, as exogenously induced by the budget questions, changes how they perceive the importance of fast earnings.

In the baseline specification, the average treatment effects are estimated via ordinary least squares as:

$Y_i = \alpha + \beta_{exp}$ Expense Discussion_i + β_{inc} Income Discussion_i + $\gamma X_i + \varepsilon_i$ (1)

where *Expense Discussion* and *Income Discussion* are indicators for random assignment to one of the treatment arms. The outcome Y_i is the relative value of the contract that pays faster, measured as the midpoint of the WTP interval recovered from the preference elicitation protocol. The estimation also controls for a set of sociodemographic and work-related covariates, X_i , which are described in table 2. The standard errors are clustered at the regional level adopted in the stratified randomization (defined as capital and non-capital areas, for each state).

The inclusion of other covariates in this estimation is justified by two reasons. First, the individual attributes we observe in the data can be structural determinants of the drivers' preferences for payment timing. In this case, they can be associated with some of the dispersion in choices, and including them as controls increases the precision of the estimates.

Second, at least one of the treatment arms is unbalanced relative to the reference group in terms of observable characteristics. If different profiles of drivers are reporting their preferences in each group, the differences in averages between treatment arms cannot be assigned to the treatment only. The introduction of the full set of covariates controls for such imbalance.

Before moving on to the results, it is useful to review what we might hope to learn from this design. A priori, the unexpected expense treatment could reinforce the perception of financial hardship and cause people to prioritize fast payment even more, especially those who already have a relatively high WTP. Alternatively, this treatment could push them to consider the long-term consequences of the trade-off more carefully, as a permanently higher income is a more effective way to cover that sort of hypothetical emergency in the future. Furthermore, if the results are driven by the specific content of the mental exercise (expenses imposing an extra burden, windfall alleviating constraints), the complementary arm with the unexpected income would flip the signs of the effect. Finally, if the effect of both treatments is simply to increase one's awareness, considering that the information recovered to answer the question sets is not too different, both treatments could lead to a similar effect, whose sign should be determined empirically.

The main experimental results are summarized in table 3. The first column reports the simple difference between the average WTP for treated and control drivers, using the midpoint of the WTP interval as the outcome. The second column reports the estimates from the regression described in equation (1), introducing the controls. Finally, the third column is an interval regression estimated using maximum likelihood, a specification that is more general because it formally incorporates the fact that the outcome is always observed between two boundaries.

	outco: WTP mi	me: idpoint	outcome: WTP interval
	Difference in Means	OLS	Interval Regression
	(1)	(2)	(3)
Treatment A:			
Unexpected expense discussion	-1.3	-1.7	-1.6
	(o.7)	(o.7)	(0.7)
Treatment B:			
Unexpected income discussion	-0.7	-1.6	-1.5
-	(o.8)	(o.7)	(o.6)
Reference level:			
Control group mean	39.9	39.9	37.4
	(o.7)	(o.7)	(o.6)
Number of observations	8,142	8,142	8,142

TABLE 3 – Effects of budget salience on the WTP for same-day remuneration

Notes: The standard errors (reported in parenthesis under the point estimate) are clustered at the regional level. For the interval regression, the estimation results are bootstrapped over 500 replications. The controls in (2) and (3) include geographical area, gender, race, age, education, household composition, work experience, previous labor market status, number of apps, vehicle ownership, work days per week, work hours per day, extra jobs, looking for another job, work income from driving, total household income, savings, and pension contribution.

The main experimental result is that both the unexpected expenses and the unexpected income discussions led to a small decrease in the importance of same-day compensation, as reported in table 3. The preferred specifications (columns 2 and 3) suggest that the average WTP for same-day remuneration is at least 1.5 percentage points lower for treated drivers, relative to those in the reference group. Still looking at the specifications that include controls, we cannot reject that the effect is statistically the same in both treatments. Notably, we also find that the effect is not homogeneous over the underlying distribution of preferences for payment timing. To investigate who is driving this result, I look at each threshold separately. Under the assumption that the ranking of preferences is stable, it is possible to stack the indifference intervals. That is, if 24% of the respondents have a WTP above 66%, and 9% have a WTP between 49% and 66%, then 33% have a WTP above 49%. This approach has the advantage of using the frontiers of the intervals as it was elicited, with no need for extra assumptions for their midpoints.

Using each possible threshold in turn, I study the level at which the effects take place, as reported at table 4. One pattern stands out: the treatments have small effects, if any, on the share of people with WTP above 6%, 11%, or 19%, but there is evidence that both treatments reduce the share of people choosing same-day remuneration against very large multipliers (1.5, 2 or 3 times) within 30 days.

Overall, drivers appear to be more likely to consider larger, deferred payments after mentally recovering their financial conditions. This result is consistent with the interpretation that the drivers in the control group are providing their first, intuitive answer to the contract choice — while treated subjects were judging the optimal balance between flexibility and long-term results with their financial context slightly more salient in their minds.

			Linea	ır Probability M	odel		
	Outcome: WTP > 3%	Outcome: WTP > 6%	Outcome: WTP > 11%	Outcome: WTP > 19%	Outcome: WTP > 32%	Outcome: WTP > 49%	Outcome: WTP > 66%
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Treatment A:							
Unexpected expense discussion	-1.9 (0.7)	-1.4 (0.8)	-0.8 (1.0)	-0.8 (1.6)	-2.2 (1.3)	-2.8 (1.0)	-2.5 (0.9)
Treatment B:			. ,				
Unexpected income discussion	0.4 (0.6)	0.3 (0.9)	-0.I (I.2)	-1.5 (1.4)	-2.6 (1.4)	-3.0 (1.0)	-2.2 (1.0)
Reference level:							
Control group mean	93.3 (0.5)	87.6 (0.7)	81.6 (0.9)	67.5 (1.1)	53.0 (1.1)	33.0 (1.0)	23.9 (1.0)
Number of observations	8,142	8,142	8,142	8,142	8,142	8,142	8,142

TABLE 4 – Average effects of budget salience on the probability of choosing a contract above a given threshold

Notes: The standard errors (reported in parenthesis under the point estimate) are clustered at the regional level. The controls include geographical area, gender, race, age, education, household composition, work experience, previous labor market status, number of apps, vehicle ownership, work days per week, work hours per day, extra jobs, looking for another job, work income from driving, total household income, savings, and pension contribution.

5.4. Potential mechanisms

From the perspective of the behavioral literature, the intervention induces a costly cognitive process that combines memory and a mental accounting exercise (*what would you do if...*). The subjects' responses retrieve particular features of their household budgets and thus provide them with an implicit reference point for the subsequent question (Gennaioli and Shleifer 2010; Shleifer 2012; Bordalo, Gennaioli, and Shleifer 2020, 2022).

In the present case, how can we make sense of the effects introduced by the treatments? Since the magnitude and signs of the effects are similar, we should look at what the treatments have in common: both require a costly information retrieval that puts the trade-off into a more complex context. In my preferred interpretation, it is precisely this common feature of both treatments that drives the modest effect I find. Relative to the control group, treated individuals were primed to actively consider their financial conditions. Once they have this information active in their minds (due to the expenses *or* the income questions), they can project themselves in the future, ponder the alternatives more carefully, and become less likely to pick the safest (but more expensive) option. The exposure to the financial question, more than its exact formulation, seems to be behind the key mechanism.

Another piece of evidence that supports this interpretation is their response time, which is precisely recorded by the survey instrument. While it is not possible to claim that individuals in both treatment groups were thinking *harder* about the trade-off, they did think *longer* than individuals in the control group. This result is consistent with treated individuals perceiving the question as a harder one, or being more careful in their choices, given the richer set of information that was made salient to them. Table 5 reports how the response time differed between treatment arms. The specification follows the baseline equation (1) closely, including the controls, except that the outcome here is the number of seconds spent on each of the three questions that make up the elicitation protocol. On average, drivers exposed to the expense discussion took 5 more seconds to complete the whole protocol, and those in the income discussion treatment took 3 seconds longer, out of an average of about 90 seconds for the control group. In both cases, the increase is most clearly identified in the third question.

This pattern is informative because the third question should, by design, offer people a trade-off closer to their indifference point. While the average time falls from the first to the third question due to the increasing familiarity with the structure of the alternatives, it does not fall as much in the treatment groups, where a share of the drivers appears to be taking the time to contemplate contracts that pay them more.

	outcome: seconds on q1	outcome: seconds on q2	outcome: seconds on q3	outcome: total seconds
	OLS	OLS	OLS	OLS
	(1)	(2)	(3)	(4)
Treatment A:				
Unexpected expense discussion	2.5	1.1	1.1	5.0
	(o.9)	(o.4)	(o.3)	(1.5)
Treatment B:				
Unexpected income discussion	0.9	0.8	1.3	3.0
	(1.1)	(0.5)	(o.3)	(1.8)
Reference level:				
Control group mean	49.9	22.5	15.8	90.1
	(1.0)	(o.4)	(0.2)	(1.5)
Number of observations	8,142	8,142	8,142	8,142

TABLE 5 – Effect of budget salience on the time to choose a contract

Notes: Response times are winsorized at 1 percent. The standard errors (reported in parenthesis under the point estimate) are clustered at the regional level. Controls include geographical area, gender, race, age, education, household composition, work experience, previous labor market status, number of apps, vehicle ownership, work days per week, work hours per day, extra jobs, looking for another job, work income from driving, total household income, savings, and pension contribution.

As a caveat, I cannot rule out that the treatments led to a modest decrease in the reported WTP each for a different reason. For instance, the mental budget under a positive shock could trigger a perception of liquidity relief, while the negative shock could lead to a budget reconsideration and increase the attractivity of larger monetary amounts. However, if one of those were indeed the key mechanism, we should expect opposite effects for each arm, since the treatments intentionally mirror the other. For this reason, I maintain as my preferred explanation that the results were driven by what the questions have in common, and not by what makes them different.

6. Robustness analysis

The main threat to the identification of the experimental effects comes from the differential attrition rate observed between the treatment arms. Individuals exposed to the unexpected expenses question were more likely to quit the survey, while those exposed to the income question were more likely to finish it.

The baseline estimation addresses this concern by including a set of sociodemographic and work-related covariates as controls in the OLS equation. In this section, I adopt doubly robust techniques to provide further evidence that the results are not induced by eventual imbalances between treatment groups (Bang and Robins 2005; Tan 2010; Wooldridge 2010). As summarized in table 6, the doubly robust estimates reinforce the finding that the increased salience of the household financial conditions induced by the expense and income questions led to a small marginal decrease in the average willingness to pay for same-day compensation. The point estimates for the doubly robust estimations are between -1.4 and -1.5 percentage points, qualitatively similar to the baseline results.

	oi WT1	utcome: P midpoint	outcome: WTP interval
	Difference in Means	Doubly Robust: IPW and Covariate Adj. via Regression	Doubly Robust: IPW and Covariate Adj. via Interval Regression
	(1)	(2)	(3)
Treatment A:			
Unexpected expense discussion	-1.3 (0.7)	-1.5 (0.7)	-1.5 (0.7)
Treatment B:			
Unexpected income discussion	-0.7 (0.7)	-1.5 (0.7)	-1.4 (0.7)
Reference level:			
Control group mean	39.9 (0.7)	40.2 (0.6)	38.9 (0.7)
Number of observations	8,142	8,142	8,142

TABLE 6 – Douł	olv robust estimation	of the effect of	budget salience	on the time to c	hoose a contract
	/		D		

Notes: The standard errors (in parenthesis) are clustered at the regional level. In (2) and (3), the standard errors also account for the estimation of the inverse probability weights (IPWs): in (2), the errors are calculated analytically; in (3), the two steps are bootstrapped over 500 replications. The additional controls used in (2) and (3), both in the regression and the propensity estimation, are the same covariates adopted in the baseline estimation.

For reference, I keep the simple difference in means in the first column. As discussed in the baseline result section, the direct comparison between the average WTP in the control group and in the treatment groups underestimates the effect of the budget discussion, particularly in the arm that discusses the use of an extra income. More importantly, columns 2 and 3 adopt the full set of controls and weight the observations by the inverse probability of being observed in the group where they are. The most conservative estimation is in column 3, as the covariate adjustment and the IPW are applied with an interval regression estimation.

	Do	ubly Robust Met	hod: Inverse Prob	ability Weight and	Covariate Adjust	ment via Regressi	on
	Outcome: WTP > 3%	Outcome: WTP > 6%	Outcome: WTP > 11%	Outcome: WTP > 19%	Outcome: WTP > 32%	Outcome: WTP > 49%	Outcome: WTP > 66%
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Treatment A:							
Unexpected expense discussion	-1.9 (0.6)	-1.3 (0.8)	-0.6 (0.9)	-0.4 (1.4)	-1.9 (1.3)	-2.7 (1.0)	-2.4 (1.0)
Treatment B:		、 ,					
Unexpected income discussion	0.5 (0.6)	0.4 (0.9)	0.0 (1.2)	-1.3 (1.3)	-2.4 (1.3)	-3.0 (1.0)	-2.2 (I.0)
Reference level:							
Control group mean	93.4 (0.4)	87.8 (0.7)	82.0 (0.8)	68.0 (1.0)	53.3 (1.1)	33.6 (0.9)	24.2 (0.9)
Number of observations	8,142	8,142	8,142	8,142	8,142	8,142	8,142

TABLE 7 – Doubly robust estimation of the effects of budget salience on the probability of choosing a contract above a given threshold

Notes: The standard errors (in parenthesis) are clustered at the regional level and account for the joint estimation of the inverse probability weights (IPWs). The additional controls, both in the regression and the propensity estimation, are the same covariates adopted in the baseline estimation.

Finally, table 7 reports the doubly robust estimates on the probability of assigning a value to the early payment option superior to each of the reference thresholds defined in the elicitation method. The same conclusion from the baseline estimation holds: the bulk of the effects come from a reduction in the share of drivers who would prefer same-day payment even against very high multipliers (that is, paying 2 or 3 times as much).

7. Concluding remarks

This chapter finds that ridesharing drivers tend to prioritize work contracts that pay faster over contracts that pay more. Such a preference is particularly strong among drivers from the poorest households, those who have little precautionary savings and no access to credit, and those who would spend their marginal dollar on food.

As a whole, this body of evidence supports the interpretation that scarcity and liquidity constraints can *by themselves* be part of the structural context that makes workers turn down offers that would pay them more. The workers who would benefit the most from higher earnings are the ones most likely to refuse them.

The simplest justification for this puzzling result is that choices that pay faster are valuable simply because they address the pressing needs of today. This chapter takes a step further and claims that the quick payment feature also compounds the benefit of flexibility in hours that is characteristic of the platform work context. If prompt payment provides a source of liquidity, prompt payment for the amount of work of one's choosing serves as insurance. This new perspective stresses how labor market arrangements can partly substitute for pure financial instruments if those are not fully available. When that happens, a work remuneration scheme is not only valued as the exchange between labor and money; it can also embed an exchange between labor and money *over time* (sooner rather than later) and *over states* (whenever is needed). Under these conditions, the same-day contract may be the most appealing alternative — even if it is an expensive one.

The conclusion that fast payment can be a safe, intuitive, automatic choice for many workers in this population is supported by the results of the experimental intervention. Simple questions about a hypothetical expense or windfall appear to remove the treated workers from the automatic setting and force them to pause and evaluate their financial conditions for a moment. The subsequent contract choices are then more reflexive, use some extra seconds of response time, and become marginally more likely to favor larger payments.

Importantly, the small magnitude of the experimental results also allows us to conclude that the very large WTP recorded for the control group is not a result of lack of attention or pure heuristics bias. Treated individuals spend significantly more time in the preference elicitation protocol and yet their average WTP reduces by no more than a few points. Whatever structural reasons explain the distribution of choices, they appear to be more relevant than the primed salience of financial circumstances.

Taking a broader perspective, the general question of the timing of the workers' paycheck has received much less attention in the labor economics literature than other job features. In this sense, this chapter claims that this dimension can be consequential and merits further research. In the context of developing countries, short payment timing is a relevant issue because it can contribute to the persistence of informal arrangements and self-employment. But workers in rich countries are not immune to similar tradeoffs. As platforms and other non-standard work arrangements become more common, payment schedules can become a more salient margin in the labor markets around the globe.

More concretely, as policymakers are actively moving to regulate platform work, this chapter invites them to consider that their relatively rapid payment is a feature appreciated by the people who have self-selected into this activity. Surprisingly, it is of primary importance for those driving full-time as well as for occasional drivers, two groups that are otherwise very different. In this sense, my implications complement the findings from Koustas (2018, 2019), who stresses how gig work can partly offset financial shocks.

The other side of this coin is that fast payment (combined with flexible labor supply) is likely one of the reasons why modern gig work can be popular while paying relatively little. The underlying risk is that it becomes a dead end: if this activity does not foster human or financial capital accumulation, people could be locked into a low-income equilibrium in which the low pay from gig work leaves them vulnerable to future shocks, which increases the insurance value of this kind of work, generating a negative feedback loop. The next step in this research agenda should be to assess if these activities lead to net welfare gains for the workers (by providing them with a viable option to mitigate shocks) or net welfare losses (by limiting their earnings in the long term).

References

- Ash, Elliott, and Stephen Hansen. 2023. "Text Algorithms in Economics." *Annual Review of Economics* 15 (1): 659–688. https://doi.org/10.1146/ annurev-economics-082222-074352.
- Bandiera, Oriana, Ahmed Elsayed, Anton Heil, and Andrea Smurra. 2022. "Economic Development and the Organisation of Labour: Evidence from the Jobs of the World Project." *Journal of the European Economic Association* 20 (6): 2226–2270. https://doi.org/10.1093/jeea/jvac056.
- Bang, Heejung, and James M. Robins. 2005. "Doubly Robust Estimation in Missing Data and Causal Inference Models." *Biometrics* 61 (4): 962–973. https://doi.org/10.1111/j.1541-0420.2005.00377.x.
- Bordalo, Pedro, Nicola Gennaioli, and Andrei Shleifer. 2020. "Memory, Attention, and Choice." *The Quarterly Journal of Economics* 135 (3): 1399–1442. https://doi.org/10.1093/qje/qjaa007.

—. 2022. "Salience." *Annual Review of Economics* 14 (1): 521–544. https: //doi.org/10.1146/annurev-economics-051520-011616.

- Brune, Lasse, Eric Chyn, and Jason Kerwin. 2021. "Pay Me Later: Savings Constraints and the Demand for Deferred Payments." *American Economic Review* 111 (7): 2179–2212. https://doi.org/10.1257/aer.20191657.
- Brune, Lasse, and Jason T. Kerwin. 2019. "Income Timing and Liquidity Constraints: Evidence from a Randomized Field Experiment." *Journal of Development Economics* 138:294–308. https://doi.org/10.1016/j.jdevec0.2019. 01.001.
- Callil, Victor, and Monise Fernandes Picanço. 2023. *Mobilidade urbana e logística de entregas: um panorama sobre o trabalho de motoristas e entregadores com aplicativos*. São Paulo: Centro Brasileiro de Análise e Planejamento. https://cebrap.org.br/wp-content/uploads/2023/05/ Amobitec12mai2023.pdf.
- Camerer, Colin, Linda Babcock, George Loewenstein, and Richard Thaler. 1997. "Labor Supply of New York City Cabdrivers: One Day at a Time." *The Quarterly Journal of Economics* 112 (2): 407–441. https://doi.org/10. 1162/003355397555244.
- Casaburi, Lorenzo, and Rocco Macchiavello. 2019. "Demand and Supply of Infrequent Payments as a Commitment Device: Evidence from Kenya." *American Economic Review* 109 (2): 523–555. https://doi.org/10.1257/aer. 20180281.

- Center for Education and Research in Innovation. 2021. "Social Security and Work on Digital Platforms." Thematic Briefing 7. São Paulo: Law School of Fundação Getulio Vargas. https://bibliotecadigital.fgv.br/dspace/ bitstream/handle/10438/30909/BT7_social_security_platforms_ingles. pdf.
- Chabris, Christopher F., David I. Laibson, and Jonathon P. Schuldt. 2016. "Intertemporal Choice." In *The New Palgrave Dictionary of Economics*, 1–8. London: Palgrave Macmillan UK. https://doi.org/10.1057/978-1-349-95121-5_1987-1.
- Chen, Kuan-Ming, Claire Ding, John A. List, and Magne Mogstad. 2020. *Reservation Wages and Workers' Valuation of Job Flexibility: Evidence from a Natural Field Experiment.* Working Paper 27807. National Bureau of Economic Research. https://www.doi.org/10.3386/w27807.
- Chen, M. Keith, Judith A. Chevalier, Peter E. Rossi, and Emily Oehlsen. 2019. "The Value of Flexible Work: Evidence from Uber Drivers." *Journal of Political Economy* 127 (6): 2735–2794. https://doi.org/10.1086/702171.
- Cohen, Jonathan, Keith Marzilli Ericson, David Laibson, and John Myles White. 2020. "Measuring Time Preferences." *Journal of Economic Literature* 58 (2): 299–347. https://doi.org/10.1257/jel.20191074.
- Cook, Cody, Rebecca Diamond, Jonathan V Hall, John A List, and Paul Oyer. 2021. "The Gender Earnings Gap in the Gig Economy: Evidence from over a Million Rideshare Drivers." *The Review of Economic Studies* 88 (5): 2210–2238. https://doi.org/10.1093/restud/rdaa081.
- Crawford, Vincent P., and Juanjuan Meng. 2011. "New York City Cab Drivers' Labor Supply Revisited: Reference-Dependent Preferences with Rational-Expectations Targets for Hours and Income." *American Economic Review* 101 (5): 1912–1932. https://doi.org/10.1257/aer.101.5.1912.
- Ericson, Keith Marzilli, and David Laibson. 2019. "Intertemporal Choice." In *Handbook of Behavioral Economics: Applications and Foundations 1*, edited by B. Douglas Bernheim, Stefano DellaVigna, and David Laibson, 2:1–67. North-Holland. https://doi.org/10.1016/bs.hesbe.2018.12.001.
- Eriksson, Tor, and Nicolai Kristensen. 2014. "Wages or Fringes? Some Evidence on Trade-Offs and Sorting." *Journal of Labor Economics* 32 (4): 899–928. https://doi.org/10.1086/676662.
- Farber, Henry S. 2008. "Reference-Dependent Preferences and Labor Supply: The Case of New York City Taxi Drivers." *The American Economic Review* 98 (3): 1069–1082. https://doi.org/10.1257/aer.98.3.1069.

- Ferrario, Beatrice, and Stefanie Stantcheva. 2022. "Eliciting People's First-Order Concerns: Text Analysis of Open-Ended Survey Questions." *AEA Papers and Proceedings* 112:163–169. https://doi.org/10.1257/pandp.20221071.
- Ferreira, Diego, Ana Cecília Kreter, Fabio Servo, Antonio Carlos Simões Florido, José Ronaldo de Castro Souza Júnior, and Guilherme Soria Bastos Filho. 2023. "Inflação de alimentos: como se comportaram os preços em 2022." Nota de Conjuntura 5. Brasília: Ipea. https://www.ipea.gov.br/cartadeco njuntura/wp-content/uploads/2023/01/230113_cc_58_nota_5_inflacao_ agro.pdf.
- Frederick, Shane, George Loewenstein, and Ted O'Donoghue. 2002. "Time Discounting and Time Preference: A Critical Review." *Journal of Economic Literature* 40 (2): 351–401. https://doi.org/10.1257/002205102320161311.
- Garin, Andrew, Emilie Jackson, Dmitri K. Koustas, and Alicia Miller. 2023. *The Evolution of Platform Gig Work, 2012-2021.* Working Paper 31273. National Bureau of Economic Research.
- Gennaioli, Nicola, and Andrei Shleifer. 2010. "What Comes to Mind." *The Quarterly Journal of Economics* 125 (4): 1399–1433. https://doi.org/10.1162/ qjec.2010.125.4.1399.
- Gentzkow, Matthew, Bryan Kelly, and Matt Taddy. 2019. "Text as Data." *Journal of Economic Literature* 57 (3): 535–574. https://doi.org/10.1257/jel. 20181020.
- Gindling, Thomas H., and David Newhouse. 2014. "Self-Employment in the Developing World." *World Development* 56:313–331. https://doi.org/10. 1016/j.worlddev.2013.03.003.
- Góes, Geraldo, Antony Firmino, and Felipe Martins. 2021. "A Gig Economy no Brasil: Uma Abordagem Inicial para o Setor de Transporte." Nota de Conjuntura 5. Brasília: Ipea. https://www.ipea.gov.br/portal/images/ stories/PDFs/conjuntura/211216_nota_5_gig_economy_brasil.pdf.
 - . 2022. "Painel da Gig Economy no Setor de Transportes do Brasil: Quem, Onde, Quantos e Quanto Ganham." Nota de Conjuntura 14. Brasília: Ipea. https://www.ipea.gov.br/portal/images/stories/PDFs/ conjuntura/220510_cc_55_nota_14_gig_economy.pdf.
- Hall, Jonathan V., and Alan B. Krueger. 2018. "An Analysis of the Labor Market for Uber's Driver-Partners in the United States." *ILR Review* 71 (3): 705– 732. https://doi.org/10.1177/0019793917717222.

- Harrison, Glenn W., and John A. List. 2004. "Field Experiments." *Journal* of *Economic Literature* 42 (4): 1009–1055. https://doi.org/10.1257/0022051043004577.
- Harvey, Melody, Cliff A. Robb, and Christopher L. Peterson. 2024. "Law and Order? Associations Between Payday Lending Prohibition and Alternative Financial Services Use by Degree of Enforcement." *Journal of Consumer Affairs* 58 (2): 538–557. https://doi.org/10.1111/joca.12583.
- Imai, Taisuke, Tom A Rutter, and Colin F Camerer. 2021. "Meta-Analysis of Present-Bias Estimation using Convex Time Budgets." *The Economic Journal* 131 (636): 1788–1814. https://doi.org/10.1093/ej/ueaa115.
- Instituto Brasileiro de Geografia e Estatística. 2023. "Pesquisa Nacional por Amostra de Domicílios Contínua." Accessed June 10, 2023. https://www. ibge.gov.br/estatisticas/sociais/trabalho/9173-pesquisa-nacional-poramostra-de-domicilios-continua-trimestral.html.
- International Labour Office. 2021. *The Role of Digital Labour Platforms in Transforming the World of Work.* 2021. Geneva: ILO.
- Koustas, Dmitri K. 2018. "Consumption Insurance and Multiple Jobs: Evidence from Rideshare Drivers." Manuscript.

——. 2019. "What Do Big Data Tell Us About Why People Take Gig Economy Jobs?" *AEA Papers and Proceedings* 109:367–371. https://doi.org/10. 1257/pandp.20191041.

- Kramer, Berber, and David Kunst. 2020. "Intertemporal Choice and Income Regularity: Non-Fungibility in the Timing of Income among Kenyan Farmers." *The Journal of Development Studies* 56 (5): 1048–1064. https: //doi.org/10.1080/00220388.2019.1632436.
- Le Barbanchon, Thomas, Roland Rathelot, and Alexandra Roulet. 2021. "Gender Differences in Job Search: Trading off Commute against Wage." *The Quarterly Journal of Economics* 136 (1): 381–426. https://doi.org/10.1093/ qje/qjaa033.
- Mas, Alexandre, and Amanda Pallais. 2017. "Valuing Alternative Work Arrangements." *American Economic Review* 107 (12): 3722–3759. https://doi.org/ 10.1257/aer.20161500.
 - . 2020. "Alternative Work Arrangements." *Annual Review of Economics* 12 (1): 631–658. https://doi.org/10.1146/annurev-economics-022020-032512.

- Matousek, Jindrich, Tomas Havranek, and Zuzana Irsova. 2021. "Individual Discount Rates: A Meta-Analysis of Experimental Evidence." *Experimental Economics* 25 (I): 318–358. https://doi.org/10.1007/s10683-021-09716-9.
- McDevitt, Ryan C., and Aaron Sojourner. 2023. "The Need for Speed: Demand, Regulation, and Welfare on the Margin of Alternative Financial Services." *The Review of Economics and Statistics*, 1–25. https://doi.org/10.1162/ rest_a_01308.
- Oh, Suanna. 2023. "Does Identity Affect Labor Supply?" *American Economic Review* 113 (8): 2055–2083. https://doi.org/10.1257/aer.20211826.
- Oyer, Paul. 2020. *The Gig Economy*. IZA World of Labor 471. Institute of Labor Economics (IZA). https://www.doi.org/10.15185/izawol.471.
- Read, Daniel. 2005. "Monetary Incentives, What Are They Good For?" *Journal* of *Economic Methodology* 12 (2): 265–276. https://doi.org/10.1080/13501780500086180.
- Scarelli, Thiago. 2022. "Occupations and Wealth in Developing Countries." *Revue d'économie du développement* 31 (2): 127–135. https://doi.org/10. 3917/edd.362.0127.
- Scarelli, Thiago, and David N. Margolis. 2023. When You Can't Afford to Wait for a Job: The Role of Time Discounting for Own-Account Workers in Developing Countries. Discussion Paper 15926. Institute of Labor Economics (IZA). https://docs.iza.org/dp15926.pdf.
- Shleifer, Andrei. 2012. "Psychologists at the Gate: A Review of Daniel Kahneman's "Thinking, Fast and Slow"." *Journal of Economic Literature* 50 (4): 1080–1091. https://doi.org/10.1257/jel.50.4.1080.
- Stegman, Michael A. 2007. "Payday Lending." *Journal of Economic Perspectives* 21 (1): 169–190. https://doi.org/10.1257/jep.21.1.169.
- Tan, Zhiqiang. 2010. "Bounded, Efficient and Doubly Robust Estimation with Inverse Weighting." *Biometrika* 97, no. 3 (September 1, 2010): 661–682. https://doi.org/10.1093/biomet/asq035.
- Thakral, Neil, and Linh T. Tô. 2021. "Daily Labor Supply and Adaptive Reference Points." *American Economic Review* 111 (8): 2417–43. https://doi.org/ 10.1257/aer.20170768.
- The Pew Charitable Trusts. 2012. *Payday Lending in America: Who Borrows, Where They Borrow, and Why.* Washington, DC. https://www.pewtrusts. org/-/media/legacy/uploadedfiles/pcs_assets/2012/pewpaydaylendingr eportpdf.pdf.

- The World Bank. 2023. *Working Without Borders: The Promise and Peril of Online Gig Work.* Washington, DC.
- Wiswall, Matthew, and Basit Zafar. 2018. "Preference for the Workplace, Investment in Human Capital, and Gender." *The Quarterly Journal of Economics* 133 (1): 457–507. https://doi.org/10.1093/qje/qjx035.
- Wooldridge, Jeffrey M. 2010. *Econometric Analysis of Cross Section and Panel Data*. 2nd ed. Cambridge, Mass.: MIT Press.

Appendix A. Additional figures and tables



FIGURE 15 – Most frequent terms mentioned by drivers when discussing how they would cover an unexpected expense (original terms in Portuguese). The word cloud depicts the 200 most frequent terms used by the ridesharing drivers who were invited to consider a situation where they would need to disburse R\$ 1,400 (US\$ 560 PPP) more than expected that week. The size and color intensity are proportional to the incidence of the term.

presentear crianca necessidade curso comemoração dentro sempre combustive vencimento imprevisto parar depositar supermercado mercado momento poupanca dois demais certeza restante acreditar boleto remedio alimentar escola pneu emergencia antecipar algo revisar aplicar levar passageiro vida milagre maravilhoso trabalhar material salvar semanal seguro mas guardar investir veiculo alegria tambem pois ganhar _{saude} risos aliviar primeiro assim comprar reserva ficar filho MUITO oha tentai roupa ano Ser app dia parte render aleluia dever valor zer voltar certo **Gioria** usar vir trocar jesus juntar coisa devolver aqua obrigado ipva financiar aplicativo viajar continuar felicidade dar comida legal futuro otimo dracas filha credito nao pai amem ter bom entrar luz pouco quitar _{mim nada} dar colocar sair rodar div passar despesa atrasar hora dificil ainda receber arrumar ate golpe gratidao mais novo origem ajuda somente mes extra melhorar mensal mae poder parcelar tirar casa estar bencao esposa hoje situacao bem **dinheiro** faturar chegar prestacao manutencao agradecer porque passeio adiantar feira reconhecimento tudo engano consertar sorte precisar familia cartao procurar pequeno debito duvidar pendencia saber deixar ^{ajeitar falta} senhor tranquilidade sobrar Surpreso realmente motorista documento verdade excelencia conseguir verificar mantimento completar gastar querer corrida bastante faculdade acertar

FIGURE 16 – Most frequent terms mentioned by drivers when discussing what they would do with an unexpected income (original terms in Portuguese). The word cloud depicts the 200 most frequent terms used by the ridesharing drivers who were invited to consider a situation where they would receive an unexpected deposit of R\$ 1,400 (US\$ 560 PPP) that week. The size and color intensity are proportional to the incidence of the term.



FIGURE 17 – Keywords from the *liquidity* discussion that distinguish the drivers with the strongest preference for same-day payment. The plot includes terms that were mentioned by more than 0.1% of the individuals and have a chi-squared statistic of at least 3.84, the critical value for 5% significance in a test with two groups. The break in the vertical axis is a reminder that all terms with a statistic in the interval [-3.84, 3.84] are omitted.



FIGURE 18 – Keywords from the *consumption* discussion that distinguish the drivers with the strongest preference for same-day payment.The plot includes terms that were mentioned by more than 0.1% of the individuals and have a chi-squared statistic of at least 3.84, the critical value for 5% significance in a test with two groups. The break in the vertical axis is a reminder that all terms with a statistic in the interval [-3.84, 3.84] are omitted.

		Ridesharing Drivers Survey							National Household Survey (PNADC)				
	All a	lrivers	Dri mai	Driver as main job		Driver as secondary job		e adult ban kforce	Male adult urban own-account workers		Male adult urban employees		
	stat.	s. e.	stat.	s. e.	stat.	s. e.	stat.	s. e.	stat.	s. e.	stat.	s. e.	
Ethinicity (share in %)													
Black	13.8	(0.30)	13.6	(0.4I)	14.5	(0.70)	11.7	(0.20)	11.4	(o.38)	12.2	(0.24)	
Mixed-race	49.6	(o.44)	49.2	(o.59)	48.1	(1.00)	44.0	(0.3I)	44.6	(0.51)	43.9	(o.38)	
White	36.6	(0.42)	37.2	(0.57)	37.3	(o.96)	44.2	(o.33)	44.0	(o.54)	43.9	(o.39)	
Age group (share in %)													
18 to 27 years old	14.3	(0.3I)	15.2	(0.42)	12.3	(o.65)	23.1	(0.23)	14.0	(o.37)	25.6	(0.30)	
28 to 37 years old	38.2	(o.43)	39.0	(0.58)	37.2	(0.96)	26.3	(0.25)	24.I	(o.47)	28.3	(o.33)	
38 to 47 years old	31.4	(0.4I)	29.9	(0.54)	34.9	(0.95)	23.9	(0.22)	25.0	(o.43)	24.0	(0.28)	
48 to 57 years old	I 2. I	(0.29)	11.8	(o.38)	12.0	(o.64)	16.8	(o.18)	20.7	(o.36)	15.2	(0.24)	
58 years old or more	4.I	(o.17)	4.I	(0.23)	3.6	(o.37)	9.9	(0.15)	16.2	(0.35)	6.9	(0.15)	
Education (share in %)													
Primary education or less	11.5	(0.28)	11.3	(o.37)	8.6	(o.56)	28.2	(0.28)	38.5	(0.50)	23.9	(o.32)	
Some high school	8.1	(0.24)	8.5	(o.33)	5.8	(o.46)	7.6	(o.14)	7.7	(o.26)	7.2	(o.17)	
High school	44.6	(o.44)	45.I	(o.59)	44.0	(o.99)	38.0	(0.29)	34.4	(o.48)	39.6	(o.37)	
Some college	20.2	(o.36)	20.9	(o.48)	20.I	(o.8o)	6.7	(0.14)	4.8	(0.21)	7.5	(0.18)	
College or above	15.5	(0.32)	14.2	(0.41)	21.4	(0.82)	19.4	(0.32)	14.6	(0.46)	21.8	(o.39)	
Household composition													
N. of adults (age 18+)	2.4	(0.01)	2.4	(0.01)	2.4	(0.02)	2.6	(0.0I)	2.5	(0.01)	2.6	(0.01)	
N. of kids (age < 18)	I.I	(0.01)	I . I	(0.01)	I . I	(0.02)	0.7	(0.01)	0.7	(0.01)	0.8	(0.01)	

TABLE 8 – Characteristics of the male ridesharing drivers in the survey and corresponding summaries for male urban adult workers

	Ridesharing Drivers Survey							National Household Survey (PNADC)				
	All d	lrivers	rivers Drive main		Driver as Driv main job second		iver as Mal dary job wor		Male adult urban own-account workers		Male adult urban employees	
	stat.	s. e.	stat.	s. e.	stat.	s. e.	stat.	s. e.	stat.	s. e.	stat.	s. e.
Work routine												
Work hours per week	53.5	(0.25)	60.7	(0.27)	33.3	(o.40)	41.6	(o.o6)	40.6	(o.14)	41.7	(0.06)
Monthly income (in R\$)												
Average work income	2,305	(15)	2,542	(18)	1,635	(24)	3,128	(35)	2,522	(41)	3,061	(36)
Average household inc. per capita	1,384	(12)	1,335	(14)	1,520	(26)	2,106	(24)	1,922	(31)	2,149	(27)
How long in this job (share in %)												
Less than 3 months	11.8	(0.3I)	9.9	(o.35)	16.0	(o.73)	10.6	(o.18)	8.6	(0.29)	I 2. I	(0.23)
3 to 6 months	9.7	(0.29)	8.9	(o.34)	12.2	(o.65)	6.0	(0.15)	4 . I	(0.29)	7.3	(o.18)
6 months to 1 year	11.5	(0.3I)	11.4	(o.38)	11.9	(o.64)	5.8	(0.13)	3.9	(0.20)	7.0	(o.17)
I to 2 years	16.6	(o.36)	16.0	(o.43)	17.9	(o.76)	10.3	(o.17)	7.1	(0.26)	I2.2	(0.23)
2 to 4 years	29.8	(o.44)	31.0	(0.55)	26.7	(o.88)	21.8	(0.22)	21.4	(0.40)	22.4	(0.28)
More than 4 years	20.5	(0.39)	22.8	(0.50)	15.3	(0.72)	45.5	(0.29)	54.9	(0.5I)	39.0	(o.35)
Social indicators (share in %)												
Contributes to a pension system	43.4	(0.55)	31.5	(o.61)	76.3	(o.93)	66.9	(0.29)	32.7	(0.50)	81.6	(0.28)
Household inc. per cap. < USD 5.5/day	II . 0	(o.33)	12.0	(o.40)	8.4	(0.57)	8.4	(o.16)	9.0	(0.29)	4.9	(o.14)
Country region (share in %)												
North	8.7	(0.25)	8.4	(o.33)	8.4	(0.55)	7.8	(0.15)	8.8	(0.25)	7.3	(o.17)
Northeast	20.5	(o.35)	20.7	(o.48)	19.6	(o.79)	21.6	(0.26)	23.0	(0.42)	20.0	(0.30)
Southeast	46.9	(o.44)	48.2	(0.59)	44.7	(o.99)	47.2	(o.36)	45.7	(0.57)	48.0	(o.46)

TABLE 8 – Characteristics of the male ridesharing drivers in the survey and corresponding summaries for male urban adult workers *(continued)*

		Ride	Drivers Sı	urvey		National Household Survey (PNADC)						
	All a	lrivers	ivers Driver as main job		Driver as secondary job		Male adult urban workforce		Male adult urban own-account workers		Male uri empl	e adult ban loyees
	stat.	s. e.	stat.	s. e.	stat.	s. e.	stat.	s. e.	stat.	s. e.	stat.	s. e.
South	13.2	(0.30)	12.4	(o.39)	15.8	(0.72)	I4.7	(0.2I)	I4.5	(0.32)	15.3	(0.27)
Central-West	10.8	(0.27)	10.3	(o.36)	11.6	(o.64)	8.7	(0.15)	7.9	(0.22)	9.3	(0.20)
<i>Survey sample</i> Number of observations	Ι3,	108	7:	,155	2,	,538	71	,858	19	,630	42,	•453

TABLE 8 – Characteristics of the male ridesharing drivers in the survey and corresponding summaries for male urban adult workers (continued)

Notes: [1] The drivers' survey was conducted by the author between the 24th and the 31st of January 2023 and its underlying population is all drivers working with a leading ridesharing company in Brazil. [2] The figures regarding to the general workforce are calculated using the microdata from Brazil's official labor survey, refer to the full year of 2022, and are weighted to be representative of the active male population above 18 years old and living in urban areas. In particular, I use the data collected by PNADC's 5th interview with the sampled households, which records household income from all sources. [4] For all variables and all subpopulations, the statistics are calculated using the available responses required for that specific item, and therefore the number of observations may vary for different attributes. The sample size for all drivers represents to the number of unique individuals who participated in the survey, while the combined number primary job drivers and secondary job drivers refer to the respondents for whom there is sufficient information for this breakdown. [4] Monetary values from PNADC are reported in January 2023 equivalent terms. [5] Work-related statistics (such as *work income, work hours* and *how long in this job*) are specific to the occupation indicated in the column. [6] The *household income per capita* is composed of all income sources from all individuals in a given household.

	All a	lrivers	Dri ma	iver as in job	Dri second	ver as lary job
	stat.	s. e.	stat.	s. e.	stat.	s. e.
Gender and ethnicity						
Male	93.2	(0.2I)	92.7	(0.30)	93.9	(o.46)
Ethinicity						
Black	13.4	(0.29)	13.1	(o.39)	14.0	(o.67)
Mixed-race	49.4	(0.42)	49.0	(0.57)	47.9	(o.96)
White	37.3	(0.4I)	37.9	(0.55)	38.1	(0.94)
Age group						
18 to 27 years old	I4.I	(o.30)	15.0	(0.4I)	I2.I	(o.63)
28 to 37 years old	38.3	(o.4)	39.1	(o.6)	37.1	(o.9)
38 to 47 years old	31.5	(o.4)	29.9	(0.5)	35.1	(o.9)
48 to 57 years old	12.2	(o.28)	12.0	(o.37)	12.0	(o.63)
58 years old or more	4.0	(o.17)	4.0	(0.22)	3.7	(0.36)
Education						
Primary education or less	11.1	(0.27)	10.9	(o.35)	8.3	(0.53)
Some high school	7.9	(0.23)	8.2	(o.31)	5.7	(0.45)
High school	44.I	(0.42)	44.7	(0.57)	43.I	(0.95)
Some college	20.7	(0.35)	21.4	(o.47)	20.5	(o.78)
College or above	16.2	(0.32)	14.8	(0.40)	22.5	(o.8o)
Household composition						
N. of adults (age 18+)	2.4	(0.01)	2.4	(0.01)	2.4	(0.02)
N. of kids (age < 18)	I.0	(0.01)	I.0	(0.01)	I.0	(0.02)
Work routine						
Work days per week	5.6	(0.01)	6.o	(0.01)	4.5	(0.03)
Work hours in a working day	9.2	(o.o3)	9.9	(0.03)	7.2	(0.06)
Work hours per week	53.0	(0.24)	60.1	(0.26)	32.9	(0.39)
Income						
Average work income	2,267	(15)	2,501	(17)	1,597	(23)
Average household inc. per capita	1,381	(12)	1,333	(13)	1,517	(25)
Household inc. per cap. < USD 5.5/day	11.3	(0.32)	12.2	(0.39)	8.4	(o.56)
How long in this job						
Less than 3 months	12.2	(0.31)	10.3	(0.35)	16.6	(0.72)
3 to 6 months	10.0	(o.28)	9.3	(o.33)	12.2	(o.63)
6 months to 1 year	11.7	(o.30)	11.7	(o.37)	12.I	(o.63)
I to 2 years	16.8	(0.35)	16.1	(0.42)	18.1	(o.74)
2 to 4 years	29.4	(0.42)	30.5	(0.52)	26.4	(o.85)
More than 4 years	19.8	(0.37)	22. I	(o.47)	I4.7	(o.68)

TABLE 9 – Descriptive statistics when ridesharing is their main or secondary job

	All c	lrivers	Dri ma	iver as in job	Dri second	ver as dary job
	stat.	<i>s. e.</i>	stat.	s. e.	stat.	s. e.
Previous status						
Inactive	4.0	(0.18)	3.6	(0.21)	4.4	(0.40)
Unemployed	29.3	(0.43)	35.6	(0.55)	12.3	(o.63)
Self-employed	22.8	(0.39)	23.0	(o.48)	21.1	(0.79)
Employee	34.7	(0.45)	28.9	(0.52)	52.2	(0.96)
Other status	9.2	(0.27)	8.9	(0.32)	9.9	(0.57)
Other jobs						
Driver only	61.6	(o.48)	85.1	(0.42)		
Driver and employee	20.8	(0.40)	3.9	(0.23)	65.5	(0.95)
Driver and self-employed	17.6	(0.38)	п.о	(0.37)	34.5	(0.95)
I ashing for a ish						
Looking for a job	0.4	(0,00)	0.5	(0 OI)	0.2	(0 OI)
	0.4	(0.00)	0.5	(0.01)	0.2	(0.01)
How many apps			<i>.</i>		0	
1 app	26.9	(0.42)	26.3	(0.50)	28.3	(0.87)
2 apps	50.9	(0.48)	50.8	(0.57)	51.1	(0.96)
3 apps	18.5	(0.37)	19.2	(0.45)	17.0	(0.72)
More than 3	3.7	(0.18)	3.8	(0.22)	3.6	(0.36)
Vehicle ownership						
Rented from friend, family	12.2	(o.31)	13.9	(o.39)	7.5	(0.51)
Rented from agency	11.9	(o.31)	13.7	(o.39)	6.9	(0.49)
Own car, still paying	56.6	(0.47)	54.7	(0.57)	61.2	(o.94)
Own car, fully paid	19.3	(o.38)	17.7	(0.43)	24.5	(o.83)
Share of work income usually saved						
Less than 10%	70.9	(0.45)	72.9	(0.52)	65.1	(o.94)
Between 10% and 25%	19.3	(0.39)	18.7	(0.45)	21.5	(o.81)
More than 25%	9.8	(0.30)	8.4	(0.32)	13.4	(0.67)
Social security						
Not currently contributing	53.I	(0.52)	63.8	(0.58)	22.5	(o.86)
Public system (as individual)	22.2	(0.43)	24.0	(0.52)	16.7	(0.77)
Public system (as employee)	15.6	(0.38)	3.5	(0.22)	50.5	(1.03)
Private system	2.3	(0.15)	1.5	(0.15)	4.5	(0.43)
Does not know	6.9	(0.26)	7.2	(0.31)	5.7	(0.48)
Country region						
North	8.8	(0.24)	8.5	(0.32)	8.3	(0.53)
Northeast	20.0	(o.34)	20.3	(0.46)	19.3	(0.76)
Southeast	46.7	(0.42)	48.0	(0.57)	44.6	(0.96)
South	13.6	(o.29)	12.8	(o.38)	16.4	(0.71)

TABLE 9 – Descriptive statistics when ridesharing is their main or secondary job (continued)

	All a	lrivers	Dri ma	iver as in job	Driver as secondary job	
	stat.	s. e.	stat.	s. e.	stat.	s. e.
Central-West	10.9	(0.26)	10.4	(0.35)	II.4	(0.61)
<i>Survey sample</i> Number of observations	14,	,265	7:	74 ^I	2,	708

TABLE 9 – Descriptive statistics when ridesharing is their main or secondary job (continued)

Notes: [1] The drivers' survey was conducted by the author between the 24th and the 31st of January 2023 and its underlying population is all drivers working with a leading ridesharing company in Brazil. [2] The figures regarding to the general workforce are calculated using the microdata from Brazil's official labor survey, refer to the full year of 2022, and are weighted to be representative of the active population above 18 years old and living in urban areas. In particular, I use the data collected by PNADC's 5th interview with the sampled households, which records household income from all sources. [3] For all variables and all subpopulations, the statistics are calculated using the available responses required for that specific item, and therefore the number of observations may vary for different attributes. The sample size for all drivers represents to the number of unique individuals who participated in the survey, while the combined number primary job drivers and secondary job drivers refer to the respondents for whom there is sufficient information for this breakdown. [4] Monetary values from PNADC are reported in January 2023 equivalent terms. [5] Work-related statistics (such as *work income, work hours* and *how long in this job*) are specific to the occupation indicated in the column. [6] The *household income per capita* is composed of all income sources from all invidividuals in a given household. [7] Non-male drivers are composed by 6.7% of female drivers and o.1% of respondents who do not identify neither as male nor female; PNADC has no comparable gender information.

	outcome: seconds on q1 IPW and Covariate Adj. via Regression (1)	outcome: seconds on q2 IPW and Covariate Adj. via Regression (2)	outcome: seconds on q3 IPW and Covariate Adj. via Regression (3)	<i>outcome:</i> <u>total seconds</u> IPW and Covariate Adj. via Regression (4)
Treatment A:				
Unexpected expense discussion	2.3	1.1	I.2	4.8
	(o.8)	(o.4)	(o.3)	(1.5)
Treatment B:				
Unexpected income discussion	0.9	0.8	1.3	3.0
	(1.0)	(o.5)	(0.3)	(1.8)
Reference level:				
Control group mean	50.1	22.5	15.9	90.5
	(1.0)	(o.4)	(0.2)	(1.4)
Number of observations	8,142	8,142	8,142	8,142

TABLE 10 – Doubly robust estimation of the effects of budget salience on the time to choose a contract

Notes: Response times are winsorized at 1 percent. The standard errors (in parenthesis) are clustered at the regional level and account for the joint estimation of the inverse probability weights (IPWs). The additional controls, both in the regression and the propensity estimation, are the same covariates adopted in the baseline estimation.

	o
12:29	.al 💻
As próximas perguntas opinião sobre modelos	pedem a sua de recebimento.
Para alguns motoristas, é por suas corridas o quant	importante receber o antes.
Outros dão preferência a mesmo que demore mais	um valor maior, para cair na conta.
Se você pudesse escolł duas opções funcionari	ıer, qual dessas a melhor para você?
Prefiro R\$ 1.54 por km sempre no dia da corri	, depositado da .
Prefiro R\$ 1.91 por km sempre 30 dias após a	, depositado corrida .
Exemplo: ao terminar um você preferiria receber R R\$ 19.1 daqui a 30 dias?	a corrida de 10 km, § 15.4 ainda hoje, ou
	\rightarrow
Desenvolvido po	r Qualtrics 🖸
Protegido por reCAPTCHA: P	

FIGURE 19 – Interface of the survey instrument

Appendix B. Survey questionnaire (English translation)

Sequence of question blocks by group

if group = {reference group}

Block 1: Geo Region Block 2: Demographics Block 3: Outcome contract choice Block 4: Making ends meet Block 5: Work and income Block 6: Open feedback

if group = {discuss income sources}

Block 1: Geo Region Block 2: Demographics Block 7: Discuss income sources Block 4: Making ends meet Block 3: Outcome contract choice Block 5: Work and income Block 6: Open feedback

if group = {discuss income uses}

Block 1: Geo Region Block 2: Demographics Block 8: Discuss income uses Block 4: Making ends meet Block 3: Outcome contract choice Block 5: Work and income Block 6: Open feedback 1.1. state

Where do you usually make your rides as an app driver?

[] Acre

- [] Alagoas
- [] Amapá
- [] Amazonas
- [] Bahia
- [] Ceará
- [] Distrito Federal
- [] Espírito Santo
- [] Goiás
- [] Maranhão
- [] Mato Grosso
- [] Mato Grosso do Sul
- [] Minas Gerais
- [] Pará
- [] Paraíba
- [] Paraná
- [] Pernambuco
- [] Piauí
- [] Rio de Janeiro
- [] Rio Grande do Norte
- [] Rio Grande do Sul
- [] Rondônia
- [] Roraima
- [] Santa Catarina
- [] São Paulo
- [] Sergipe
- [] Tocantins

1.2. capital

In the capital or in other areas?

- [] {State capital} and surrounding areas
- [] Another city of {State}

2.1. gender

Your gender?

- [] Male
- [] Female
- [] Other
- [] Prefer not to answer

2.2. race

With which of these options do you identify yourself most?

- [] White
- [] Brown
- [] Black
- [] Indigenous
- [] Asian

2.3. age

How old are you?

- [] Between 18 and 22 years old
- [] Between 23 and 27 years old
- [] Between 28 and 32 years old
- [] Between 33 and 37 years old
- [] Between 38 and 42 years old
- [] Between 43 and 47 years old
- [] Between 48 and 52 years old
- [] Between 53 and 57 years old
- [] Between 58 and 62 years old
- [] Between 63 and 67 years old
- [] 68 years old or more

2.4. schooling

What is your schooling degree?

- [] No schooling
- [] Primary school, incomplete
- [] Primary school, complete
- [] Secondary school, incomplete
- [] Secondary school, complete
- [] Bachelor's degree (college), incomplete
- [] Bachelor's degree (college), complete
- [] Graduate school, incomplete
- [] Graduate school, complete

2.5. hh_adults

How many adults (18 years or older) live in your household, including you?

[] I adult (just me)

[] 2 adults

- [] 3 adults
- [] 4 adults

[] 5 adults

[] 6 adults or more

2.6. hh_kids

How many children and teenagers (up to 18 years old) live in your house-hold?

- [] no children / teenager
- [] I children / teenager
- [] 2 childrens / teenagers
- [] 3 childrens / teenagers
- [] 4 childrens / teenagers
- [] 5 childrens / teenagers
- [] 6 childrens / teenagers or more

The next questions ask you for your opinion on payment models.

For some drivers, it's important to get paid as soon as possible. Others prefer a higher amount, even if it takes longer to arrive in their account.

3.1. s_or_l

If you could choose, which of these two options would work best for you?

[] I prefer R\$ {reference rate} per km, always deposited on the day of the ride.
[] I prefer R\$ {reference rate × 1.24} per km, always deposited 30 days after the ride.

Example: at the end of a 10 km ride, would you prefer to receive R {*reference rate* × 10} *today, or* R {*reference fee* × 1.24 × 10} *in 30 days*?

if $s_or_l == \{ on the day of the ride \}$

3.2. sas_or_las

And in this case, which of these two options would work best for you? [] I prefer R\$ {reference rate} per km, always deposited on the day of the ride. [] I prefer R\$ {reference rate × 1.96} per km, always deposited 30 days after the ride.

Example: at the end of a 10 km ride, would you prefer to receive R {*reference rate* × 10} *today, or* R {*reference fee* × 1.96 × 10} *in 30 days*?

if $s_or_l == \{30 \text{ days after the ride}\}$

3.3. sal_or_lal

And in this case, which of these two options would work best for you? [] I prefer R\$ {reference rate} per km, always deposited on the day of the ride. [] I prefer R\$ {reference rate × 1.06} per km, always deposited 30 days after the ride.

Example: at the end of a 10 km ride, would you prefer to receive R {*reference rate* × 10} *today, or* R {*reference fee* × 1.06 × 10} *in 30 days*?

if sas_or_las == {on the day of the ride}

3.4. sass_or_lass

And in this case, which of these two options would work best for you? [] I prefer R\$ {reference rate} per km, always deposited on the day of the ride. [] I prefer R\$ {reference rate × 2.92} per km, always deposited 30 days after the ride.

Example: at the end of a 10 km ride, would you prefer to receive R {*reference rate* × 10} *today, or* R {*reference fee* × 2.92 × 10} *in 30 days*?

if sas_or_las == $\{30 \text{ days after the ride}\}$

3.5. sasl_or_lasl

And in this case, which of these two options would work best for you? [] I prefer R\$ {reference rate} per km, always deposited on the day of the ride. [] I prefer R\$ {reference rate × 1.48} per km, always deposited 30 days after the ride.

Example: at the end of a 10 km ride, would you prefer to receive R {*reference rate* × 10} *today, or* R {*reference fee* × 1.48 × 10} *in 30 days*?

if sal_or_lal == {on the day of the ride}

3.6. sals_or_lals

And in this case, which of these two options would work best for you? [] I prefer R\$ {reference rate} per km, always deposited on the day of the ride. [] I prefer R\$ {reference rate × 1.12} per km, always deposited 30 days after the ride.

Example: at the end of a 10 km ride, would you prefer to receive R\$ {reference rate × 10} today, or R\$ {reference fee × 1.12 × 10} in 30 days?

if sal_or_lal == $\{30 \text{ days after the ride}\}$

3.7. sall_or_lall

And in this case, which of these two options would work best for you? [] I prefer R\$ {reference rate} per km, always deposited on the day of the ride. [] I prefer R\$ {reference rate × 1.03} per km, always deposited 30 days after the ride.

Example: at the end of a 10 km ride, would you prefer to receive R {*reference rate* × 10} *today, or* R {*reference fee* × 1.03 × 10} *in 30 days*?

Block 4: Making Ends Meet

4.1. making_ends_meet

Overall, how easy is it to make ends meet in your household?

[] Very easy

[] Easy

[] Relatively easy

[] Neither easy nor hard

[] Relatively hard

[] Hard

[] Very hard

Block 5: Work and Income

5.1. how_long_app

How long have you been working as an app driver?

If you have stopped this activity for over 3 months, consider only the period since you resumed it.

- [] Less than a month
- [] Between one and 3 months
- [] Between 3 and 6 months
- [] Between 6 months and one year
- [] Between one and 2 years
- [] Between 2 and 4 years
- [] More than 4 years

5.2. previous_state

What was your situation the month before you started (or resumed) working with ridesharing apps?

[] Student

- [] Unemployed
- [] Self-employed worker
- [] Full-time employee
- [] Part-time employee
- [] On leave due to long illness or other incapacity
- [] Taking care of the household full-time
- [] Retired
- [] Other

if previous_state == {Unemployed}

5.3. previous_state_unemp

In the month before you started (or resumed) working with ridesharing apps, were you looking for a job?

[] Yes [] No

if previous_state == {Full-time employee} or {Part-time employee}

5.4. previous_state_emp

In the month before you started (or resumed) working with ridesharing apps, were you a formal employee? [] Yes [] No
if previous_state == {Self-employed worker}

5.5. previous_state_oaw

In the month before you started (or resumed) working with ridesharing apps, did you have a formal registration as a self-employed worker? [] Yes

[]No

5.6. main_reasons

At that moment, what led you to start (or resume) this activity?

Taking into account the other activities I could do, I decided to be a driver because...

[] it paid me more than my other options.

[] it was more enjoyable than my other options.

[] it was easier to conciliate with my personal life.

[] allowed me to work according to my current needs.

[] I could secure some income quickly.

[] driving is my best professional skill.

[] there were no other options available at that moment.

[] I had other reasons: [_____]

5.7. how_many_apps

How many apps do you currently work with?

[] I [] 2 [] 3 [] more than 3

5.8. working_vehicle

Which option best describes your current work vehicle?

[] Own car, paid for

[] Own car, still paying for it

[] Car rented from an agency

[] Car rented from a friend/family member

[] Car rented via the app's partners

[] Borrowed vehicle

5.9. work_days_per_week

How many days a week do you usually work as a driver, on average?

[] Less than I day per week

- [] I day per week
- [] 2 days per week
- [] 3 days per week
- [] 4 days per week
- [] 5 days per week
- [] 6 days per week
- [] 7 days per week

5.10. work_hours_per_day

How many hours do you usually drive during a work day, on average?

- [] Less than 1 hour
- [] I hours
- [] 2 hours
- [] 3 hours
- ... [] 22 hours [] 23 hours
- [] 24 hours

5.11. other_jobs

Do you currently have any paid activities other than driving?

- [] Yes, other activities as a self-employed worker
- [] Yes, as a full-time employee
- [] Yes, as a part-time employee
- [] No, driving is currently my only paid activity

if other_jobs == {Yes, other activities as a self-employed worker}

5.12. other_jobs_oaw

In this other activity, do you have a formal registration as a self-employed worker?

- [] Yes
- []No

if other_jobs == {Sim, empregado(a) tempo integral} or {Sim, empregado(a) tempo parcial}

5.13. other_jobs_emp

In this other employment, are you a formal employee?

[] Yes

[]No

if other_jobs \neq {No, driving is currently my only paid activity}

5.14. main_or_second_inc

The driving activity is currently...

[] my main income source.

[] a secondary income source.

5.15. looking_for_a_job

Are you currently looking for a job?

[] Yes [] No

5.16. driver_income

What is your monthly net income as a driver, in approximate terms?

Consider the income available to you after paying for fuel and the other car costs.

[] Less than R\$ 500 per month

- [] R\$ 500 to R\$ 1,000 per month
- [] R\$ 1,000 to R\$ 1,500 per month
- [] R\$ 1,500 to R\$ 2,000 per month
- [] R\$ 2,000 to R\$ 2,500 per month
- [] R\$ 2,500 to R\$ 3,000 per month
- [] R\$ 3,000 to R\$ 3,500 per month
- [] R\$ 3,500 to R\$ 4,000 per month
- [] R\$ 4,000 to R\$ 5,000 per month
- [] R\$ 5,000 to R\$ 6,000 per month
- $[\]$ R\$ 6,000 to R\$ 7,000 per month
- [] R\$ 7,000 to R\$ 8,000 per month
- [] R\$ 8,000 to R\$ 10,000 per month
- [] Mais de R\$ 10,000 per month

5.17. hh_income

What is the total income in your household, in approximate terms?

Consider all the incomes of all the residents, including your net income as a driver and your other activities.

[] Less than R\$ 500 per month

- [] R\$ 500 to R\$ 1,000 per month
- [] R\$ 1,000 to R\$ 2,000 per month
- [] R\$ 2,000 to R\$ 3,000 per month
- [] R\$ 3,000 to R\$ 4,000 per month
- [] R\$ 4,000 to R\$ 5,000 per month
- [] R\$ 5,000 to R\$ 6,000 per month
- [] R\$ 6,000 to R\$ 7,000 per month
- [] R\$ 7,000 to R\$ 8,000 per month
- [] R\$ 8,000 to R\$ 10,000 per month
- [] R\$ 10,000 to R\$ 12,000 per month
- [] R\$ 12,000 to R\$ 15,000 per month
- [] Mais de R\$ 15,000 per month

5.18. savings

How much of your net income as a driver do you usually save at the end of the month?

- [] Nearly nothing (0% to 10%)
- [] A small share of it (10% to 25%)
- [] A good share of it (25% to 40%)
- [] Approximately half of it (40% to 60%)
- [] A large share of it (60% to 75%)
- [] Most of it (75% to 90%)
- [] Nearly all of it (90% to 100%)

if savings > 10%

5.19. savings_destination

What are the main purposes of these reserves?

- [] Possible emergencies related to work (my car broke down, I got sick, etc.)
- [] Possible domestic emergencies (home, family)
- [] Future professional training
- [] A new business
- [] Leisure and holidays
- [] Saving for retirement
- [] Buying a specific good (house, car, appliance, etc.)
- [] Specific personal event (birthday, wedding, etc.)
- [] My reserves do not have a specific purpose
- [] Other objectives: [_____]

5.20. pension

Do you currently contribute to a pension?

[] I contribute to the public system as a self-employed

- [] I contribute to the public system as an employee
- [] I contribute to a private pension scheme
- [] I don't contribute to any pensions at the moment
- [] I wouldn't know how to answer that

if pension == {I don't contribute to any pensions at the moment}

5.21. why_no_pension

What are the main reasons you don't contribute to a pension at the moment?

[] I would like to, but I don't know how it works

- [] I would like to, but the contributions are too expensive
- [] I would like to, but there is no money left for that
- [] I am saving by myself with what is left at the end of the month
- [] I am saving by myself with a fixed monthly amount
- [] The returns are too low, it is not worth it
- [] It is too early to think about that
- [] I do not trust the pension system
- [] I am already retired
- [] Other reasons: [_____]

Block 6: Open Feedback

6.1. feedback

Thank you very much for your attention!

If you like, you can leave a comment on the survey.

In general, what did you think of the questions? Did you have any difficulties or discomfort?

[------]

Now, let's consider a hypothetical situation.

Imagine that you have received news of a domestic emergency (an urgent home repair, or a health treatment that cannot wait).

Because of this, you'll have to disburse R\$1,400 more than expected this week.

7.1. priming_income_sources_word

What's the first word that comes to mind in a situation like this? [_____]

7.2. priming_income_sources_descr

In practice, how would you cover this unforeseen expense of R\$ 1,400 right now?

Think about the situation and describe your options in a few words.

[_____]

Block 8: Discuss Income Uses

Now, let's consider a hypothetical situation.

Imagine you received news of a surprise payment (the result of a lottery or an unexpected refund, for example).

Because of this, you will receive an extra deposit of R\$ 1,400 this week.

8.1. priming_income_uses_word

What's the first word that comes to mind in a situation like this?

[-----]

8.2. priming_income_uses_descr

In practice, what would you do with this unexpected gain of R\$ 1,400 right now?

Think about the situation and describe your options in a few words.

[_____]